Misallocation and Capital Market Integration: Evidence From India^{*}

Natalie Bau[†] Adrien Matray[‡]

Abstract

We show that foreign capital liberalization reduces capital misallocation and increases aggregate productivity in India. The staggered liberalization of access to foreign capital across disaggregated industries allows us to identify changes in firms' input wedges, overcoming major challenges in the measurement of the effects of changing misallocation. For domestic firms with initially high marginal revenue products of capital (MRPK), liberalization increases revenues by 23%, physical capital by 53%, wage bills by 28%, and reduces MRPK by 33% relative to low MRPK firms. There are no effects on low MRPK firms. The effects of liberalization are largest in areas with less developed local banking sectors, indicating that foreign capital partially substitutes for an efficient banking sector. Finally, we develop a novel method to use natural experiments to bound the effect of changes in misallocation on treated industries' aggregate productivity. Treated industries' Solow residual increases by 3-16%.

^{*}We are particularly indebted to David Baqaee and Chenzi Xu. We thank Dave Donaldson, Emmanuel Farhi, Pete Klenow, Karthik Muralidharan, Diego Restuccia, Richard Rogerson, Martin Rotemberg, Chad Syverson, Christopher Udry, Liliana Varela, as well as conference and seminar participants at the Stanford King Center Conference on Firms, Trade, and Development, CEPR Macroeconomics and Growth Meetings, CIFAR IOG meetings, EPED, NBER SI, the Online International Finance and Macro Seminar, Columbia, Toulouse School of Economics, INSEAD, CREST, University of Paris-Dauphine, Georgetown, the World Bank, Dartmouth, UToronto, UCLA, UCSD, Guelph, and USC-Marshall Business School for helpful comments and discussions. Carl Kontz, Palermo Penano, Brian Pustilnik, Derek Wenning, and Mengbo Zhang provided exceptional research assistance. We are also grateful to the International Growth Centre, the Julis-Rabinowitz Center for Public Policies, and the Griswold Center of Economic Policy Studies (Princeton), which funded this project.

[†]UCLA, NBER, and CEPR. (email: nbau@ucla.edu)

[‡]Princeton and CEPR. (email: amatray@princeton.edu)

1 Introduction

The misallocation of resources across competing uses is a leading explanation for economic disparities across countries. However, identifying policies that can affect misallocation, and quantifying their aggregate effects remains a major challenge. There are at least two reasons for this.

On the measurement side, it is common to attribute all - or much of - the cross-sectional dispersion in the observed marginal returns to firms' inputs to misallocation. This creates upward bias in measures of misallocation and can contaminate estimates of differences in allocative efficiency across countries or over time.¹

On the policy side, even if one were able to fully correct for mismeasurement and quantify the effect of changes in misallocation on aggregate productivity, the specific sources of misallocation are difficult to identify from aggregate comparisons.² This leaves policymakers with limited information about what levers to pull to reduce misallocation. In low-income countries, where there are likely to be large firm-level frictions in the allocation of resources, understanding which policies reduce misallocation could provide policymakers with powerful tools to foster economic growth.

An unusual natural experiment in India allows us to make progress on both the measurement and the policy fronts, providing some of the first evidence on a policy tool that can be used to reduce misallocation. Over the 2000s, India introduced the automatic approval of foreign direct investments up to at least 51% of domestic firms' equity, potentially reducing capital market frictions. Using the staggered introduction of the policy across industries, we implement a differencein-differences framework to estimate the effects of this foreign capital liberalization on the misallocation of capital across firms. In the absence of a natural experiment, the measurement of changes in misallocation would be contaminated by

^{1.} Upward bias can come, for example, from measurement error (Bils, Klenow, and Ruane, 2018; Rotemberg and White, 2017; Gollin and Udry, 2021), model misspecification (Haltiwanger, Kulick, and Syverson, 2018; Nishida, Petrin, Rotemberg, and White, 2017), volatility of productivity paired with the costly adjustment of inputs (Asker, Collard-Wexler, and De Loecker, 2014; Gollin and Udry, 2021), unobserved heterogeneity in technology (Gollin and Udry, 2021), and informational frictions and uncertainty (David, Hopenhayn, and Venkateswaran, 2016; David and Venkateswaran, 2019).

^{2.} To quantify the overall degree of misallocation, the literature usually compares outcomes such as the distribution of marginal revenue products across units of production after controlling for different observable characteristics and attributes the residual dispersion to misallocation. Since this method of quantifying misallocation typically does not show which characteristics causally affect the residual dispersion in marginal products, it is mostly silent on what policies would be required to reduce misallocation in low-income countries.

measurement error and other (unobserved) shocks. However, in this setting, the natural experiment allows us to isolate changes in inputs and the change in the marginal revenue product of capital due to the policy. The inclusion of various controls, such as firm, year, and 5-digit industry-year fixed effects, controls for many sources of unobserved heterogeneity that could otherwise bias measurement.

A priori, the effect of opening-up to foreign capital on allocative efficiency is unclear. On the one hand, in low-income countries, where formal credit markets are limited, opening up to foreign capital markets might reduce funding constraints if foreign investors have better screening technologies or are not bound by domestic historical, political, or regulatory constraints.³ On the other hand, foreign investors may also be worse at processing and monitoring soft information, particularly in low-income countries, worsening the allocation of capital.⁴

We find that the liberalization of foreign capital reduces capital misallocation by increasing capital for the firms with the highest marginal revenue returns to capital prior to the reform. We then develop a method, based on the theoretical results of Petrin and Levinsohn (2012) and Baqaee and Farhi (2019), to translate our quasi-experimental microeconomic estimates into lower and upper bound measures of the effect of the policy on the treated industries' Solow residual (a proxy for these industries' aggregate productivity). Our proposed method uses exogenous variation to generate a lower bound for the aggregate effect of changing misallocation under relatively weak identifying assumptions, without relying on cross-sectional dispersion in marginal revenue products.

To measure the effects of the reform, we collected data on industry-level liberalization episodes in 2001 and 2006. Combining this policy variation with a panel of large and medium-sized Indian firms over the period 1995–2015, we investigate whether the reform reduced misallocation by testing whether the policy had differential effects depending on firms' ex-ante marginal revenue products of capital (henceforth "MRPK"). By exploiting within-industry variation in firms' MRPK dispersion, this empirical strategy allows us to evaluate whether the policy reduced misallocation while controlling for differential trends across industries over time. Thus, estimating whether the policy reduced misallocation does not require deregulation to be random, nor for firms to have similar levels of pre-reform covariates, or even for treated and untreated industries to be on the same trends prior to

^{3.} See Townsend (1994), Banerjee, Duflo, Glennerster, and Kinnan (2015), Burgess and Pande (2005), and Cole (2009) for examples of domestic frictions in financing.

^{4.} In the context of foreign banks' behavior in low-income countries, several studies have found that foreign banks mainly lend to large domestic firms, thereby potentially increasing credit constraints for local firms (e.g., Detragiache, Tressel, and Gupta, 2008).

the reforms. It only requires that high MRPK firms are not growing relatively more quickly than low MRPK firms in treated vs untreated industries prior to the reform, an assumption that we provide visual evidence for using event studies. In our most stringent specifications, we can account for any unobserved shocks or differences in time trends at the disaggregated industry, state, and size quartile levels.

We find that, in response to the policy, high MRPK firms in deregulated industries increase their physical capital by 53%, revenues by 23%, wage bills by 28%, and reduce their MRPK by 33%, relative to low MRPK firms. In contrast, low MRPK firms are not affected. Since high MRPK firms had more than 170% higher MRPK than low MRPK firms, the micro-estimates imply that the policy reduces dispersion in MRPK. Event study graphs confirm that these effects are not driven by the gap between high and low MRPK firms evolving differently within treated industries relative to untreated industries. They also provide evidence that the reduction in misallocation is not due to mean reversion.

To better understand the mechanism underlying these results, we exploit geographic variation in local access to credit prior to the reform. We find that the effects of liberalization on misallocation are largest in areas where the local banking sector was less developed. This is consistent with the hypothesis that foreign investors can reduce misallocation by standing in for, and competing with, local credit markets.

We next explore the effect of the reform on firms' products, including product portfolio, prices, and quantities. This is made possible by a rare feature of our firm-level data set: detailed data on each firm's product-mix, product-level output, and prices. Since a reduction in distortions on input prices should reduce marginal costs for affected firms, firms may pass some of these gains onto consumers via lower prices. Depending on the degree of pass-through, the change in the price could be greater than or less than the change in the marginal cost. We find that the reform differentially reduced prices for high MRPK firms in treated industries by 12% but had no significant effect on the prices of low MRPK firms. Additionally, high MRPK firms in treated industries increase the number of products in their portfolio, in part by introducing more new products.

The liberalization policy may have had broader effects than reducing firms' wedges on capital inputs. If firms need to borrow to pay workers, relaxing financial constraints can also affect labor misallocation.⁵ Motivated by this possibility, we examine the effect of the policy on labor misallocation. Analogous to our approach

^{5.} For more discussion of this mechanism, see Fonseca and Doornik (2021) in Brazil.

for capital, we estimate the policy's differential effect on firms with high marginal revenue products of labor (henceforth, "MRPL"). We find wage bills only increased for firms with high MRPL. For these firms, relative to low MRPL firms, wage bills increased by 24%, and MRPL fell by 28%. Since high MRPL firms had at least two times higher levels of MRPL prior to the treatment in treated industries, disperion in MRPL and hence, labor misallocation also fell.

Finally, combining production function parameter estimates with reduced-form estimates of the policy effect, we generate bounds on the effect of the liberalization on the treated industries' Solow residual. As a lower bound, the treated industries' Solow residual increased by 3%. Accounting for the cumulative effects of the policy over time raises this number to 6%. Even at a lower bound, the policy had economically meaningful aggregate effects. In contrast, if we infer baseline wedges from the pre-treatment cross-sectional data, the upper bound effect is 16%.

The paper is organized as follows. The remainder of the introduction discusses the related literature. Section 2 provides a brief conceptual framework for understanding misallocation and introduces the expression we will use for aggregation. Section 3 describes the data and the context of the policy change. Section 4 discusses our reduced-form empirical strategy. Section 5 reports our estimates of the average effect of the foreign capital liberalization policy and its heterogeneous effects on firms with high and low MRPK. It also replicates the analysis for firms that have high and low MRPL to test whether the policy also reduced labor misallocation. Section 6 describes the aggregation strategy and reports estimated bounds on the foreign capital liberalization policies' effect on the Solow residual for treated industries. Section 7 concludes.

Related Literature. This paper contributes to two main literatures. First, it contributes to the literature quantifying the importance of misallocation for aggregate outcomes (e.g. Restuccia and Rogerson, 2008; Hsieh and Klenow, 2009; Bartelsman, Haltiwanger, and Scarpetta, 2013; Baqaee and Farhi, 2020; David and Venkateswaran, 2019; Sraer and Thesmar, 2020), particularly in the context of developing countries (e.g., Guner, Ventura, and Xu, 2008; Banerjee and Moll, 2010).⁶ Second, it contributes to the literature on the effects of financial frictions and misallocation (e.g., Buera, Kaboski, and Shin; 2011; Midrigan and Xu, 2014; Moll, 2014; Hombert and Matray, 2016; Kehrig and Vincent, 2019; Bai, Carvalho, and Phillips, 2018; Delatte, Matray, and Pinardon Touati, 2019).

Regarding the misallocation literature, much of the previous work has focused

^{6.} A survey of this literature can be found in Restuccia and Rogerson (2017).

on measuring the effect of all sources of misallocation on aggregate output by exploiting cross-sectional dispersion in marginal revenue products. The principal advantage of this "indirect approach" (Restuccia and Rogerson, 2017) is that it allows for the estimation of the overall cost of misallocation without identifying the underlying sources of the distortions, even if the sources are not observable to researchers. However, in this approach, model misspecification and measurement error can inflate estimates of misallocation and bias estimates of the effects of changing misallocation.

We make three contributions to this literature. First, since we exploit a liberalization episode that affected only certain industries, we can estimate the effect of deregulation on misallocation using milder identification assumptions than the literature that depends on cross-sectional or cross-country variation. Our difference-in-differences strategy only requires that measurement error or other unobserved attributes are uncorrelated with the policy change to identify *changes* in input wedges. Second, our approach isolates the changes in distortions produced by a specific policy, foreign capital liberalization.⁷ This allows us to isolate the effect of access to the foreign equity market, holding constant access to the foreign debt market and other macroeconomic determinants that might affect the cost of capital differentially for different firms. Third, we show how our natural experiment estimates can be used to compute aggregate effects of reducing misallocation that are less vulnerable to inflation due to measurement error or model mis-specification. In so doing, we develop a method that can be applied in other contexts by researchers studying misallocation.

By developing a general method that exploits a natural experiment to identify changes in misallocation and quantify their effects on aggregate productivity, we also relate to Sraer and Thesmar (2020). Sraer and Thesmar (2020) develop a sufficient statistics approach that uses estimates from natural experiments to calculate the counterfactual effects of scaling-up a policy to the entire economy. This is fundamentally different from the object we bound — the aggregate effect of the policy that was actually enacted — which can be bounded with relatively few assumptions about firms' production functions and interactions.

In terms of capital account liberalization, this paper relates most closely to

^{7.} In the context of India, several recent papers have studied specific characteristics of the Indian economy that might explain the high degree of misallocation observed in the country: the role of property rights and contract enforcement (Bloom et al., 2013); land regulation (Duranton, Ghani, Goswani, and Kerr, 2017); industrial licensing (Chari, 2011); privatization (e.g., Dinc and Gupta, 2011); reservation laws (Garcia-Santana and Pijoan-Mas, 2014; Martin, Nataraj, and Harrison, 2017; Boehm, Dhingra, and Morrow, 2019; Rotemberg, 2019); electricity shortages (Allcott, Collard-Wexler, and Connell, 2016), and labor regulation (Amirapu and Gechter, 2019).

a recent strand of this literature that has explored how increased foreign financial flows affect domestic firms' productivity, sectoral misallocation, and welfare (e.g., Alfaro, Chanda, Kalemli-Ozcan, and Sayek, 2004; Gopinath, Kalemli-Özcan, Karabarbounis, and Villegas-Sanchez, 2017; Varela, 2017; Xu, 2020). We add to this literature in two ways. First, while much of the previous literature exploits country-level variation in access to foreign investment, this paper exploits variation across industries over time within the same country. This allows us to hold the institutional setting constant, which is important since institutional differences affect cross-country comparisons. Second, since the Indian deregulation only affected foreign investment in equity, it allows us to cleanly isolate the effect of foreign investment in equity on misallocation, holding fixed access to foreign $debt.^8$

Our results contrast with those of Gopinath, Kalemli-Özcan, Karabarbounis, and Villegas-Sanchez (2017), who find that increased access to foreign debt increased misallocation in Spain. However, these findings are not mutually exclusive. First, it is unclear whether expansions in equity and debt will have similar effects. Equity encourages investors to invest in firms with high upside potential, while debt may encourage investment in "safe" firms with high levels of collateral. Second, the baseline levels of misallocation and financial development are surely different in India from those in an OECD country, and this context is likely to be important for determining the size and direction of the effects of increased financial integration (e.g., Varela, 2017).

More broadly, our paper relates to the literature that studies misallocation in developing countries, which has particularly focused on agriculture. This literature has studied the possibility that missing markets (land, insurance, credit) may give rise to misallocation, which in turn has implications for the distributional impact of productivity shocks and technology shocks (Gollin and Udry, 2021), risk sharing (Townsend, 1994), the impact of microcredit (Kaboski and Townsend, 2011), property rights (e.g., Adamopoulos, Brandt, Leight, and Restuccia, 2017), and the farm size distribution (Restuccia, Yang, and Zhu, 2008; Adamopoulos and Restuccia, 2014; Restuccia and Rogerson, 2017). This literature was among the first to use panel data to improve upon cross-sectional analyses of input and output dispersion, which may be contaminated by unobserved heterogeneity (e.g. Udry, 1996; LaFave and Thomas, 2016; Gollin and Udry, 2021).

^{8.} In contrast, Varela (2017) studies the deregulation of capital controls in Hungary, in a context where foreign capital was already integrated and was not affected by the policy.

2 Conceptual Framework

2.1 Misallocation and Reduced-Form Predictions

We follow standard practice in the literature and model misallocation as wedges on the prices of inputs. Intuitively, the wedges can be thought of as explicit or implicit taxes that implement a given (potentially inefficient) allocation in the decentralized Arrow-Debreu-McKenzie economy. Thus, the allocative price paid by a firm *i* for an input *x* is $(1 + \tilde{\tau}_i^x)p^x$, where $x \in \{K, L, M\}$ and *K*, *L*, and *M* denote capital, labor, and materials, respectively. The observed price of input *x* is p^x , and $\tilde{\tau}_i^x$ is the additional wedge a firm pays for the input over the observed price. The wedge $\tilde{\tau}_i^x$ can be negative, indicating that a firm is subsidized, or positive, indicating that the firm pays a tax relative to the observed price. A single-product firm's profit function is

$$\pi_i = p_i f_i(K_i, L_i, M_i) - \sum_{x \in \{K, L, M\}} (1 + \tilde{\tau}_i^x) p^x x_i$$

where $f_i(K_i, L_i, M_i)$ is the firm's production function, which exhibits diminishing marginal returns in each input.

A cost-minimizing firm will consume an input x_i until that input's marginal revenue returns $p_i \partial f_i(K_i, L_i, M_i) / \partial x_i$ are equal to the cost

$$p_i \frac{\partial f_i(K_i, L_i, M_i)}{\partial x_i} = \mu_i (1 + \tilde{\tau}_i^x) p^x$$

where μ_i is the mark-up or output wedge.⁹ Then, define the combined wedge $1 + \tau_i^x = \mu_i(1 + \tilde{\tau}_i^x)$. The marginal revenue product of input x is proportional to the (combined) wedge τ_i^x . Therefore, firms with higher combined input wedges τ_i^x (capital, labor or any other) will have higher marginal revenue products on this input (henceforth, "MRPX").

We now generate partial equilibrium predictions that we can use to test for a reduction in misallocation in the data. A decrease in the misallocation of input x occurs when the wedge τ_i^x declines for a firm whose wedge is high relative to other firms. A decline in the wedges of firms with relatively high initial τ_i^x will

^{9.} Technically, if firm *i* has pricing power, then the marginal revenue product of an input x (MRPX) is better defined as $p_i \partial f_i(K_i, L_i, M_i) / \partial x_i + \partial p_i / \partial x_i f_i(K_i, L_i, M_i)$ rather than $p_i \frac{\partial f_i(K_i, L_i, M_i)}{\partial x_i}$. This is because a change in x both directly affects a firm's output and (if it has pricing power) its price. However, in the misallocation literature, MRPX typically refers to $p_i \frac{\partial f_i(K_i, L_i, M_i)}{\partial x_i}$ because it is dispersion in this value that causes misallocation. Thus, we use this definition of MRPX at the cost of abusing terminology.

have several effects. First, since τ_i^x falls, the measured MRPX should also fall for these firms. Second, firms with high wedges will increase their use of x. Finally, the increase in input x (say capital) will increase the marginal revenue products of the other inputs, which will incentivize firms to also increase their demand for these other inputs (e.g. labor). As a result of higher input use, these firms will produce more and earn higher revenues. Thus, if the policy reduces capital misallocation by reducing the wedges of firms with high τ_i^k , we should expect to find that the policy increases capital, labor, and sales, and decreases MRPK for firms with ex-ante high values of MRPK.

2.2 Framework for Quantifying Effects on Solow Residual

To quantify the effects of reducing misallocation on treated industries' aggregate productivity, following much of the literature, we proxy for changes in aggregate productivity with changes in the Solow residual, which measures the net output growth minus the net input growth. Thus, denoting the Solow residual for a sector of interest I as $Solow_I$,

$$\Delta Solow_I = \Delta Net \ Output_I - \Delta Net \ Input_I. \tag{1}$$

Net output growth is the change in the treated firms' output net the outputs reused as inputs by treated firms. Net input growth is the change in the inputs used by treated firms net of the inputs that are produced by treated firms. Let net output of good i be $c_i = y_i - \sum_{j \in I} y_{ji}$, where y_i is the output of firm i and y_{ji} are the inputs used by firm j of the output of i. The change in the treated firms' net output is defined as $\Delta C_I = \sum_{i \in I} p_i \Delta c_i$. This is the total change in net quantities valued using fixed prices. The Solow residual in discrete time is then

$$\Delta Solow_I = \Delta \log C_I - \sum_{j \notin I} \frac{\sum_{i \in I} p_j y_{ij}}{\sum_{i \in I} p_i c_i} \Delta \log \sum_{i \in I} y_{ij}.$$
 (2)

The summation $\sum_{j \notin I}$ sums over firms that supply intermediate goods to firms in the treated industries but are not themselves treated, while the summation $\sum_{i \in I}$ sums over firms in the treated industries. Thus, $\Delta \log C_I$ measures the change in output due to the policy (differencing out outputs that are re-used as inputs), while the latter term in equation (2) subtracts out changes in inputs purchased from outside the treated industries. Intuitively, as shown in equation (1), the Solow residual measures the change in output valued using current market prices and differences out the growth in inputs valued using those same prices. Thus, in an accounting sense, it controls for input growth due to the policy.

In general, as demonstrated by Petrin and Levinsohn (2012) and Baqaee and Farhi (2019), a first order approximation of the change in the Solow residual of the set of treated firms in I over time is given by:

$$\Delta Solow_{I,t} \approx \sum_{i \in I} \lambda_i \Delta \log A_i + \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_i \alpha_i^x \frac{\tau_i^x}{1 + \tau_i^x} \Delta \log x_i \tag{3}$$

where λ_i is the ratio of firm *i*'s sales to treated industry *I*'s net output, $\Delta \log A_i$ is the change in total factor productivity (TFPQ), α_i^x is the output elasticity with respect to x, τ_i^x is the level of firm-specific input wedges prior to the policy change, and $\Delta \log x_i$ is the change in the log input x consumed by firm *i*, which itself is endogenous to A_i . This expression allows us to convert firm-level effects, which are in different units depending on the goods being produced, into aggregate effects.¹⁰ A derivation of this expression is provided in Appendix A. We show that this expression does not require any assumptions about returns to scale, cross-good aggregation, or the shape of input-output networks. As we will explain in Section 6, equation (3) will allow us to exploit our reduced-form estimates to put bounds on the aggregate effect of the policy change on the treated industries' Solow residual.

3 Data and Policy Change

3.1 Indian Foreign Investment Liberalization

Following its independence, India became a closed, socialist economy, and most sectors were heavily regulated.¹¹ However, in 1991, India experienced a severe balance of payments crisis, and in June 1991, a new government was elected. Under pressure from the IMF, the World Bank, and the Asian Development Bank, which offered funding, the Indian government engaged in a series of structural reforms. These reforms led India to become more open and market-oriented. In addition to initiating foreign capital reforms in more than one-third of the

^{10.} Note that increasing firms' capital inputs does not strictly increase the Solow residual in equation (3) for two reasons. First, for a fixed amount of total capital in the economy, increasing capital for a low MPRK firm reduces capital for a high MPRK firm, reducing the Solow residual. Second, if a firm faces a negative wedge (it is subsidized), $\frac{\tau_i^x}{1+\tau_i^x}$ will be negative, and the positive change in inputs will be multiplied by a negative value.

^{11.} See Panagariya (2008) for a thorough review of the Indian growth experience and government policies.

manufacturing sector in this period, India also liberalized trade (e.g. Topalova and Khandelwal, 2011; Goldberg, Khandelwal, Pavcnik, and Topalova, 2010) and dismantled extensive licensing requirements (e.g. Aghion, Burgess, Redding, and Zilibotti, 2008; Chari, 2011).

Before 1991, most industries were regulated by the Foreign Exchange Regulation Act (1973), which required every instance of foreign investment to be individually approved by the government, and foreign ownership rates were restricted to below 40% for each firm in most industries. With the establishment of the initial liberalization reform in 1991, foreign investment up to 51% of equity in certain industries became automatically approved.¹² In the following years, different industries liberalized at different times, with each liberalization increasing the cap on foreign investment and allowing for automatic approval.

We study the effects of financial liberalization episodes that occurred after 2000, well after the main period of reform in the 1990s. This is both due to data availability, as described below, and to avoid conflating the effects of the financial liberalization reforms with other ongoing reforms. The timing of these later reforms seems to reflect the political climate in India. The initial wave of reform in 1991 was highly controversial, and the government adopted a more gradual approach to limit political push-back (Singh, 2005). However, by the early 2000s, both major political parties actively supported pursuing foreign investment, and additional reforms were further instigated by declines in total foreign direct investment in India in the late 1990s (Singh, 2005). While FDI can take a variety of forms in India, from joint partnerships with technological transfers to the direct purchase of equity in Indian firms (with little additional training or technological investment), the latter form has been much more common following the reforms (Beena et al., 2004).

We collected data on the timing of disaggregated industry-level policy changes from different editions of the *Handbook of Industrial Policy and Statistics*. We match this data to industries at the 5-digit NIC level. An industry is coded as having been treated if a policy change occurred that allowed automatic approval and/or increased the cap on investments to at least 51% of capital (though, in some cases, the maximum is higher). We then merge this data at the industry-level with the firm-level dataset described below.

^{12.} This policy is described by Topalova (2007) and Sivadasan (2009).

3.2 Firm and Product-Level Data

Our firm-level data comes from the Prowess database compiled by the Centre for Monitoring the Indian Economy (CMIE) and includes all publicly traded firms, as well as a large number of private firms. Unlike the Annual Survey of Industries (ASI), which is the other main source of information used to study dynamics in the Indian manufacturing sector, Prowess is a firm-level panel dataset. Therefore, the data is particularly well-suited for examining how firms adjust over time in reaction to policy changes. The dataset contains information from the income statements and balance sheets of companies comprising more than 70% of the economic activity in the organized industrial sector of India and 75% of all corporate taxes collected by the Government of India. It is thus representative of large and medium-sized Indian firms. We retrieve yearly information about sales, capital stock (measured as tangible, physical assets), consumption of raw materials and energy, and compensation of employees for each firm. Unfortunately, Prowess does not contain information on number of employees.

To estimate the effect of the reform on prices, we take advantage of one rare feature in firm-level datasets that is available in Prowess: the dataset reports both total product sales and total quantity sold at the firm-product level, allowing us to compute unit prices and quantities. This unusual feature is due to the fact that Indian firms are required by the 1956 Companies Act to disclose product-level information on capacities, production, and sales in their annual reports.¹³ The definition of a product is based on Prowess's internal product classification, which is in turn based on India's national industrial classification (NIC) and contains 1,400 distinct products. Using this information, we can calculate the unit-level price for each product, which we define as total unit sales over total unit quantity. This allows us to also construct a separate panel of product-level output and prices.¹⁴

3.3 Main Combined Datasets

To arrive at our final datasets for analysis, we merge the firm-level and productlevel panel data with the industry-level policy data.

^{13.} A detailed discussion of the data can be found in Goldberg, Khandelwal, Pavcnik, and Topalova (2010).

^{14.} One limitation of this dataset is that firms choose which type of units to report, and units are not always standardized across firms or within-firms over time. Thus, when we want to analyze the effects of policy changes on prices/output and there is not enough information to reconcile changes in unit types within a firm-product over time, we drop the set of observations associated with a firm-product. We omit 2% of observations.

As is common in the literature, we restrict our analysis to manufacturing firms. We further restrict the sample to observations from the period between 1995 and 2015. Restricting the sample to 1995–2015 has two advantages. First, focusing on this later period avoids potential bias from other liberalization reforms during the early-1990s, the main Indian liberalization period. While 45% of manufacturing firms in the data are in industries that liberalized at some point, by restricting our sample to observations after 1995, we only exploit policy variation from the 10% of manufacturing firms who experienced foreign capital liberalizations in the 2000s. Second, although Prowess technically starts in 1988, its coverage in the first few years was limited and grew substantially over time. In 1988, Prowess only included 735 manufacturing firms total, but it had grown to 3,652 firms by the beginning of our study period in 1995. In contrast, from 1995 onward, during our study period, the coverage of the database is more stable, with similar numbers of firms observed across subsequent years (3,664 firms observed in 1996, 3,470 in 1997, and 3,614 in 1998).¹⁵

Additionally, to allow for a longer pre-policy period over which to calculate MRPK and classify MRPK as high or low, as described below, we drop a very small number of observations that experienced a liberalization in 1998. This amounts to 104 total firm-year observations (roughly 4–5 per year) or 0.26% of the sample. Appendix Table A1 provides a list of the different industries in the manufacturing sector affected by the deregulation during the remaining sample. As the table shows, after dropping the 1998 liberalization, the only remaining liberalization episodes occurred in 2001 and 2006.

Following other work with manufacturing data in India (e.g., Hsieh and Klenow (2009); Rotemberg and White (2017)), to reduce noise from misentry in the data, for our main analyses, we also drop observations where there are extreme negative year-to-year fluctuations in revenues (declines greater than 85%). This reduces our sample size by 8%, but – as we will show in our robustness section – our main results are virtually unchanged when we do not make this restriction.

Finally, we restrict the sample to the set of firms for which we can compute marginal revenue products of capital and labor (MRPK and MRPL) prior to the earliest policy change in 2001. These pre-treatment measures are needed to estimate the effects of the policy on misallocation. Thus, we restrict the sample to firms observed before 2001 with non-missing, positive data on both assets and

^{15.} This likely reflects the fact that the first wave of liberalizing reforms also standardized financial reporting in the mid-1990s.

	Treated			Control		
	Mean	s.d.	Median	Mean	s.d.	Median
Foreign (%)	4.1	19.7	0.0	7.0	25.6	0.0
State Owned (%)	3.7	18.8	0.0	2.0	14.1	0.0
Firm Age	26.4	18.9	21.0	26.7	18.9	21.0
Gross Fixed Assets	25.5	132.4	3.5	18.2	49.9	3.1
Sales/Revenues	64.3	274.4	12.3	57.8	146.6	11.9
Wages	3.6	19.7	0.6	4.2	9.8	0.8
MRPK (Revenue/K)	7.5	89.4	3.1	9.3	71.6	3.2
Observations	56,861			$6,\!288$		

Table 1: Summary Statistics

This table reports summary statistics for the manufacturing firms appearing in the CMIE Provess dataset from 1995 to 2015. An observation is at the firm-year level. Firms' capital, income, salaries, and revenues are measured in millions of USD and are deflated with industry deflators.

sales.¹⁶ These restrictions leave us with an unbalanced panel of 5,013 distinct firms across 343 distinct 5-digit industries, for a total of 63,149 observations.

Table 1 documents summary statistics for the final firm-level sample used in our analysis. As the table shows, the typical firm in our analysis is a domestic firm, and few firms are state-owned. In our final sample, 10% of firms are in industries that experienced a policy change between 1995 and 2015.

3.4 Supplementary Data Sources

While most of our analyses are conducted using only Prowess, we also supplement it with a variety of additional data sources. We outline each briefly in this section.

We use two additional datasets to provide evidence that the policy had first stage effects on access to foreign capital: (1) data on foreign loans (available after 2004) scrapped from the Reserve Bank of India's website and (2) CapEx data on the timing and ownership of large capital projects in India. To ensure our results are robust to the inclusion of small establishments, we also evaluate the effects of the policy using the Annual Survey of Industries, a survey of a representative cross-section of Indian manufacturing establishments conducted by the Indian government. Additionally, to evaluate whether the policy had heterogeneous effects in states with less-developed banking systems, we collected data from the Reserve

^{16.} This is the minimal requirement to calculate MRPK. As we document in the next subsection, we exploit the fact that, under Cobb-Douglas production functions, sales divided by capital will be proportional to MRPK within an industry, as long as α_j^k is the same for all firms in industry j.

Bank of India at the state-level for each of the pre-reform years (1995–2000) on the total credit of all scheduled commercial banks. Finally, for our robustness analyses, we supplement the Prowess data with data on Indian dereservation policies¹⁷ (the removal of protective laws for small-scale industries) and input and output tariff data,¹⁸ both at the 5-digit industry-level.

4 Empirical Strategy

4.1 Measurement: MRPK and TFPQ

To determine whether foreign capital liberalization reduces misallocation, we follow the predictions in our conceptual framework and test if the reform has a differential effect on firms with high and low MRPK. Below, we describe the method used to measure firms' MRPK.

As is standard in the production function estimation literature,¹⁹ we assume that firms have Cobb-Douglas revenue production functions:

$$Revenue_{ijt} = TFPR_{ijt}K_{ijt}^{\alpha_j^k}L_{ijt}^{\alpha_j^l}M_{ijt}^{\alpha_j^m}$$

$$\tag{4}$$

where *i* denotes a firm, *j* denotes an industry, and *t* denotes a year. $Revenue_{ijt}$, K_{ijt} , L_{ijt} , and M_{ijt} are measures of sales, capital, the wage bill, and materials, and $TFPR_{ijt}$ is the firm-specific unobserved revenue productivity. Throughout this paper, capital is measured as the total value of tangible, physical assets.

To estimate MRPK, we take advantage of the fact that, under the revenue Cobb-Douglas production function, $MRPK = \frac{\partial Revenue_{it}}{\partial K_{it}} = \alpha_j^k \frac{Revenue_{it}}{K_{it}}$. Thus, $\frac{Revenue_{it}}{K_{it}}$ provides a within-industry measure of MRPK, under the assumption that all firms in an industry share the same α_j^k . To determine whether firms had a high or low MRPK prior to the reform, we average each firm's measures of MRPK over 1995–2000 (the last year prior to the first policy change). We then classify a firm as high MRPK if its average MRPK is above the 4-digit industry-level median.

In addition to measuring MRPK, we also create a measure of TFPQ as a

^{17.} Garcia-Santana and Pijoan-Mas (2014), Martin, Nataraj, and Harrison (2017), Boehm, Dhingra, and Morrow (2019), and Rotemberg (2019) describe these laws and estimate their consequences. We thank Boehm, Dhingra, and Morrow for generously sharing the data with us.

^{18.} India experienced a massive reduction in its trade tariffs in the 1990s, as has been studied by Topalova and Khandelwal (2011) and Goldberg, Khandelwal, Pavcnik, and Topalova (2010). We would like to thank Johanes Boehm for generously sharing his tariff measure with us.

^{19.} Duranton, Ghani, Goswani, and Kerr (2017) describe a variety of methods used to estimate production functions and the revenue returns to capital and labor.

proxy for firm-level productivity. We implement the Levinsohn and Petrin (2003) method (henceforth "LP"), using the GMM estimation proposed by Wooldridge (2009), to estimate the parameters of revenue production functions at the 2-digit industry-level.^{20,21} The LP method estimates the parameters of the production function using a control function approach, where materials are assumed to be increasing in a firm's unobserved productivity conditional on capital. This identifying assumption does not require that capital or labor are not misallocated — the key sources of misallocation that we study in this paper — but does assume away misallocation of materials. For the production function estimation, we measure inputs and revenues with deflated Ruppee amounts, so that $Revenue_{iit}$ is proxied with deflated sales.²² The revenue production function allows us to calculate revenue total factor productivity, TFPR. Using the product data, which measures unit prices, we calculate $\log TFPQ = \log TFPR - \log \tilde{p}$, where \tilde{p} is the sales share weighted average of the prices of a firm's products. By estimating the effect of the reform on TFPQ, we can examine whether foreign capital liberalization affects within-firm productivity as well as misallocation. The sample size for which TFPQ is available is much smaller (43,791 firm-year observations), as calculating this measure requires data on all firm inputs, as well as price data. Thus, we view our within-firm level productivity results as more exploratory than our main misallocation results.

4.2 Main Specification: Heterogeneous Effects

To measure the effect of liberalization on the allocation of resources across firms within industries, we estimate the following equation:

$$Outcome_{ijt} = \beta_1 \ Reform_{jt} + \beta_2 Reform_{jt} \times I_i^{High \ MRPK} + \Gamma \mathbf{X}_{it} + \theta_i + \delta_t + \epsilon_{ijt}$$

$$\tag{5}$$

^{20.} In principle, we could use the quantity data to directly estimate quantity production functions, but in practice, relying on this data greatly reduces the sample size available for estimation.

^{21.} One concern is that multi-product firms produce goods in multiple industries, leading to bias when we estimate production function parameters at the industry-level. We use the firmlevel industry identifiers provided by Prowess to assign firms to industries (Prowess provides a single industry value for each firm), and this issue is partially mitigated by the fact that subsidiaries of large conglomerates in different industries appear as different observations in the data.

^{22.} We use deflators for India made available by Allcott, Collard-Wexler, and Connell (2016) for the period 1995–2012, and we extended the price series to 2015. Revenue is deflated using three-digit commodity price deflators. The materials deflators are measures of the average output deflator of a given industry's suppliers using the 1993-4 input-output table. The capital deflator is obtained using an implied national deflator.

where *i* denotes a firm, *j* denotes an industry, *t* denotes a year, and $Outcome_{ijt}$ is the outcome variable of interest, consisting of the logs of physical capital, the total wage bill, sales, and MRPK. $Reform_{jt}$ is an indicator variable equal to one if foreign investment has been liberalized in industry *j*, and $I_i^{High MRPK}$ is an indicator variable equal to 1 if a firm has a high pre-reform MRPK according to our measure defined in Section 4.1. \mathbf{X}_{it} consists of firm age and firm pre-treatment size-by-year fixed effects,²³ so that β_1 and β_2 are identified by comparing two firms with the same age and within the same size bin. In a robustness check, we show that our main results are robust to including a more parsimonious set of controls. θ_i and δ_t are firm and year fixed effects respectively. δ_t controls for aggregate fluctuations, while θ_i removes time invariant unobserved firm-level heterogeneity, which may bias estimates of the MRPK dispersion.²⁴ Standard errors are two-way clustered at the 4-digit industry and year level to account for any serial correlation that might bias our standard errors downward.²⁵

The coefficient of interest is β_2 , which captures the differential effect of the reform on ex-ante high MRPK firms relative to low MRPK firms. $\beta_2 > 0$ implies that the dependent variable increases for high MRPK firms relative to low MRPK firms in industries that have opened up to foreign capital relative to industries that have not. β_1 measures changes in low MRPK firms' outcomes, and $\beta_1 + \beta_2$ measures total changes in high MRPK firms' outcomes.

4.3 Identification

Below, we discuss the extent to which our empirical strategy is vulnerable to three potential sources of bias: (1) non-random assignment of treatment status across firms, (2) the endogeneity of foreign equity flows, and (3) measurement error in MRPK. We also emphasize that the policy can reduce misallocation even if foreign investors do not directly identify and invest in high MRPK firms.

Selection of treated firms. One natural concern is that firms in industries that are liberalized are different from firms in industries that are not. As long

^{23.} Firm size is defined as fixed effects for the within 2-digit industry quartiles of firms' average pre-treatment capital.

^{24.} As previously discussed, cross-sectional measures of MRPK are likely to be inflated by measurement error. Indeed, if we calculated the level of capital misallocation using cross-sectional data, a standard approach would be to use an estimate of the variance of MRPK as a proxy for the dispersion of the wedges. This estimate would sum over both the variance of the wedges and the variance of measurement error, leading to inflated estimates of the dispersion of the wedges.

^{25.} Our treatment variable is coded at the 5-digit industry-level, but we cluster at the 4-digit level to account for possible correlations across more closely related industries.

as these differences are time-invariant, this selection is fully accounted for by firm fixed effects (θ_i). Similarly, firm fixed effects account for any time invariant differences, observed or unobserved, between high and low MRPK firms. Thus, our specification does *not* require that the reform was randomly allocated, nor does it require that firms must have the same pre-treatment characteristics.

A classic difference-in-differences strategy requires that treated firms have the same time trends as untreated firms in the absence of the reform. However, because we exploit differences within deregulated industries to estimate β_2 , our key parameter for evaluating the change in misallocation, our identification assumption for β_2 does not require "strict exogeneity." Even if treated and control industries are on different trends (i.e., $E[\epsilon_{ijt} \ Reform_{jt}] \neq 0$), we can still identify β_2 by controlling directly for $Reform_{jt}$ or more conservatively, for 5-digit industry-year fixed effects.

 β_2 will be unbiased as long as the *difference* in outcomes between ex-ante high and low MRPK firms in treated industries would have evolved the same way as in control industries in the absence of the reform. Note that this does *not* imply that high MRPK firms have to have the same time trends as low MRPK firms, but simply that the *difference* in trends between low and high MRPK is the same in treated and control industries.

While identifying β_2 does not require the standard difference-in-differences assumption that trends are the same across treatment and control industries for the average firm, when we turn to the aggregation exercise in Section 6, we will use our estimates of both β_1 and β_2 . In contrast to solely identifying β_2 , identifying β_1 requires that trends are parallel between treated and untreated industries. We provide support for this assumption in two ways. First, we visually assess whether there are parallel pre-trends between treated and untreated industries in an event study figure. Second, we show that our estimates of both β_1 and β_2 are insensitive to the inclusion of additional controls for differential time trends at the firm and industry-level.

Endogeneity of foreign equity flows. While it is likely that, within an industry, foreign capital is targeted towards specific firms, we do not use observed variation in foreign capital in our regressions. Instead, we exploit an exogenous shifter to the amount of foreign capital an industry can receive. Therefore, to be unbiased, β_1 and β_2 do not require that foreign capital is allocated randomly across firms in treated industries. As long as the differential time trends assumptions discussed above are not violated, our approach delivers valid estimates of the effect of liberalizing industry-level access to foreign capital.

Measurement error in MRPK. Measurement error should have little effect on our estimates if it is either firm-specific and time-invariant or time-variant but common across firms in a given year. Firm fixed effects and year fixed effects account for systematic measurement error at the firm and year level.

On the left side of the equation, as is well-known in the econometrics literature, classical measurement error (i.e. error independent of the latent true variable) in the outcome variable will not bias the point estimates. On the right side, idiosyncratic measurement error in MRPK may bias our estimate of β_2 if it leads to error in the coding of $I_i^{High MRPK}$. This measurement error would lead some firms that are actually high MRPK to be coded as low MRPK, while some low MRPK firms will be coded as high MRPK. As long as the true effect of the policy is to reduce MRPK more for ex-ante high MRPK firms, misclassification will lead to attenuation bias. Since β_2 captures the change in high MRPK firms' capital wedges, this would lead us to *underestimate* the change in these firms' wedges due to the policy. However, non-classical measurement error could still bias our results in the other direction. We return to this issue in Section 5, when we show that our reduced-form estimates are not sensitive to winzorizing extreme values.

Investors allocate FDI in response to characteristics besides MRPK. Our test of the effect of the policy on misallocation does not require that foreign investors knowingly invest more in high MRPK firms or even that foreign investment specifically increases for high MRPK firms. It could be, for example, that foreign investors invest more in large, well-established low MRPK firms, reducing these firms' need for external financing. This would reduce their demand for capital from the domestic financial sector, freeing up resources that could then be redirected to smaller, high MRPK firms. In this case, greater access to capital for high MRPK firms would be a "by-product" of greater access to capital at the industry level.²⁶ Regardless of whether foreign investors can identify and directly target high MRPK firms or not, foreign capital liberalization policies reduce misallocation if they lead to a relative increase in capital for ex-ante high MRPK firms.

^{26.} It is impossible to test this directly because India does not have a credit registry. However, there is recent evidence that a similar reallocation of capital happened in Europe when the ECB introduced its program purchasing corporate bonds. Arce, Mayordomo, and Gimeno (2020) report that large firms with access to the bond markets issued more bonds, leading to a drop in the demand for bank loans by bond issuers. This led banks to increase their supply of credit to smaller, non-bond issuing firms that were constrained prior to the ECB intervention.

Nonetheless, in Section 5.1, we evaluate whether foreign investment appears to directly target high MRPK firms.

5 Results

5.1 First Stage: Access to Foreign Capital

Before estimating the effect of the policies on misallocation, we provide industrylevel evidence that foreign capital flows increased in treated industries. Additionally, using two different proxies for accessing foreign capital, we provide evidence that increased access to foreign capital was concentrated in ex-ante high MRPK firms.

Industry-level evidence. First, we obtain evidence on industry-level FDI flows by exploiting the "Equity Composition" module of Prowess, which was collected from 2001 onward. This module, which focuses on publicly-listed firms, covers less than a quarter of firms in our sample. Therefore, these data cannot be analyzed at the firm-level in our difference-in-differences framework. Instead, we aggregate the total FDI flows to listed firms at the industry-year level. In Figure A1, we report the overall amount of foreign equity when we bundle industries by year of FDI regulation and normalize the flows to their initial, 2001 levels. While we cannot observe if there is a trend-break for industries treated in 2001, since there is no pre-period, the figure suggests that the growth in FDI accelerates over the post-period (the red line) relative to all the sectors whose status did not change over the period (the blue line). The effect is even more striking for industries that liberalized in 2006 (the green line), as there is a clear trend break after 2006.

Firm-level evidence. To analyze the reform's firm-level effects, we use two proxies for access to foreign capital. First, we take advantage of information on another form of foreign financing: foreign debt. Second, we observe whether a firm implements a large capital project that is at least partially financed by foreign investors.

Since 2004, all Indian firms must report any foreign loans to the Reserve Bank of India (RBI) in a database called "External Commercial Borrowings." We scrapped this data from the RBI website and match all foreign loan information to Prowess using the name of the company. We then construct two outcome measures: (1) an indicator variable equal to one if a firm has ever accessed the foreign market, and (2) the inverse hyperbolic sine of the cumulative amount of foreign loans.²⁷ Acknowledging the limitation that we only have 2 pre-treatment years and can only analyze the effects of the 2006 reform, we estimate our differencein-differences with heterogeneous effects specification. Table 2 reports the results. The odd columns report the average effect of the reform, and the even columns report the heterogeneous effects by MRPK. The estimates indicate that ex-ante high MRPK firms differentially increase any access of foreign debt by 6 percentage points and increase their total foreign debt by 96%.

For our second measure, we use another dataset maintained by CMIE – CapEx - to further measure increases in foreign capital flows. CapEx compiles data on all projects that entail a capital expenditure of 10 million Rupees or more (roughly 135,000 USD) in India.²⁸ The data include a company identifier that can be matched to Prowess and an ownership variable indicating whether the project is foreign-owned or not. We emphasize that these data certainly undercount foreign capital flows, since a project is unlikely to be marked as foreign-owned if it is minority foreign-funded, since not all FDI will result in capital projects, and since the data do not include smaller capital projects. We report the same specifications as above with an indicator variable equal to one if a firm has any foreign-owned project (columns 5–6) and the inverse hyperbolic sine of cumulative spending on foreign projects (columns 7–8) as our outcome variables. We find similar patterns to the ones for foreign debt. Ex-ante high MRPK firms are more likely to have any foreign project post-treatment and have larger spending on foreign-owned projects. Based on our two proxies for foreign capital flows, we conclude that ex-ante high MRPK firms receive more foreign capital as a result of the reforms.

5.2 Average Effects

We estimate the effect of the reform on the average firm's financial outcomes by removing the interaction term $Reform_{jt} \times I_i^{High MRPK}$ from equation (5). Table 3 reports the results. The estimates indicate that the liberalization policy has positive effects on firm investments. For the average firm, capital increases by 32%

^{27.} The inverse hyperbolic sine transformation of the log function (e.g., Burbidge, Magee, and Robb, 1988; MacKinnon and Lonnie, 1990) is defined as: $log[X + (X^2 + 1)^{1/2}]$. Except for very small values of X, the *ihs* is approximately equal to log(2X) or log(2) + log(X) and can be interpreted in exactly the same way as a standard logarithmic dependent variable. However, unlike a log transformation, the inverse hyperbolic sine is defined at zero and is not overly sensitive to jump around zero, unlike the more classic log(x + 1) transformation.

^{28.} Examples of projects include the creation, expansion, or renovation and modernization of factories or retail establishments.

Dependent Variable	Any Foreign Debt		Ihs(Foreign Debt)		Any Foreign Project		Ihs(Foreign Spending)	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
$Reform_{jt}$	$0.030 \\ (0.019)$	-0.005 (0.013)	$\begin{array}{c} 0.451 \\ (0.345) \end{array}$	-0.078 (0.213)	$0.004 \\ (0.003)$	$0.0002 \\ (0.003)$	$\begin{array}{c} 0.029\\ (0.022) \end{array}$	0.001 (0.020)
$Reform_{jt} \times I_i^{High MRPK}$		0.064^{**} (0.025)		$\begin{array}{c} 0.963^{***} \\ (0.326) \end{array}$		0.007^{***} (0.002)		0.050^{**} (0.018)
Fixed Effects								
Firm	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$61,\!156$	$61,\!156$	$61,\!156$	$61,\!156$	$61,\!156$	$61,\!156$	$61,\!156$	$61,\!156$

Table 2: Effect of the Reform on Proxies for Foreign Direct Investment

This table reports the effect of the reform on access to foreign debt and the cost of capital projects financed by foreign investors. In columns 1–4, we use the universe of foreign loans since 2004 tracked by the RBI to study if firms issue any foreign debt (columns 1–2) or to estimate the total change in foreign debt (columns 3–4). Due to the large number of zeros, we use the inverse hyperbolic sine transformation (denoted Ihs(x)) in columns 3–4 and 7–8. In columns 5–8, we use the dataset CapEx, compiled by CMIE, which contains data on all large capital projects in India. The outcomes are an indicator variable for whether a firm is associated with a foreign-owned project (columns 5–6), and the *Ihs* of the total spending on foreign-owned projects associated with the firm (columns 7–8). Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10, 5, and 1% statistical significance respectively.

(column 2), while average MRPK declines by 18.7% (column 4), both significant at the 5% level. The point estimates for the total wage bill and revenues are also positive, albeit not significant at conventional levels (the wage bill is borderline significant with p = .12). Figure 1 plots the event study graph for the average effects on capital, reporting the yearly effect of belonging to a treated industry before and after the reform, including the same controls as in Table 3. Here, 0 is normalized to be the year before a reform. Consistent with the absence of differential pre-trends, we see that there is no effect of belonging to a treated industry before the reform took place.

5.3 Differential Effects by Ex-ante MRPK

Baseline specification. Table 4 reports the estimates of the heterogeneous effects of the policy from equation (5), our main estimating equation. Following the liberalization, high MRPK firms generate higher revenues by 23% (column 1) relative to low MRPK firms. This is made possible by the fact that these firms invest more, with their physical capital differentially increasing by 53% (column 2). Higher investment does not crowd-out labor. High MRPK firms also experience a relative increase in their wage bills by 28%, suggesting that there may be important

Dependent Variable	Revenues Capital		Wages	MRPK	
	(1)	(2)	(3)	(4)	
$Reform_{jt}$	$0.108 \\ (0.094)$	$\begin{array}{c} 0.318^{**} \\ (0.118) \end{array}$	$0.159 \\ (0.098)$	-0.187^{**} (0.088)	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	$58,\!391$	60,096	59,162	57,017	

Table 3: Average Effect of the Foreign Capital Liberalization

All dependent variables are in logs. $Reform_{jt}$ is an indicator variable equal to one if the industry had liberalized access to the international capital market in or before year t and zero otherwise. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. In column 4, MRPK is computed using Revenue/K as a proxy for the marginal revenue product of capital. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10, 5, and 1% statistical significance respectively.

Figure 1: Event Study Graph for the Average Effect of Foreign Capital Liberalization on Physical Capital



This figure reports the event study graph for the average effect of the liberalization on firms' physical capital. The dependent variable is in logs. The reform is normalized to take place in year 1. Each dot is the coefficient on the interaction between being observed t years after the reform and being in a treated industry. The confidence interval is at the 95% level.

complementarities between capital and labor. We will further explore whether the reform also reduced labor misallocation in Section 5.6. Among the ex-ante high MRPK firms, the policy relatively reduced MRPK by 33% (column 4).

The coefficient on the variable $Reform_{jt} \times I_i^{High MRPK}$ measures the marginal effect of the reform on firms' outcomes for high MRPK firms *relative* to low MRPK firms. Following the reform, high MRPK firms invest more relative to low MRPK

Dependent Variable	Revenues Capital		Wages	MRPK	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_i^{High MRPK}$	0.226^{***} (0.076)	0.527^{***} (0.075)	0.280^{***} (0.058)	-0.326^{***} (0.107)	
$Reform_{jt}$	-0.024 (0.125)	$\begin{array}{c} 0.017 \\ (0.089) \end{array}$	-0.002 (0.097)	$0.004 \\ (0.102)$	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	$58,\!391$	60,096	59,162	$57,\!017$	

Table 4: Heterogeneous Effects of Foreign Capital Liberalization by Firms' Exante MRPK

All dependent variables are in logs. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995–2000 is above the 4-digit industry median. MRPK is estimated with the Revenue/K method. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

firms, and their marginal revenue return on capital falls. Given that, prior to the reform, high MRPK firms had a MRPK more than twice as high as low MRPK firms, this implies that the reform shrank but did not fully eliminate the gap between high and low MRPK firms' MPRK. Thus, the reform led to a decline in the dispersion of MRPK (and a substantial reduction in misallocation).

We also use the same empirical strategy to examine whether the composition of capital changed heterogeneously as a result of the reform. Table A2 reports the results when the outcome variables are the share of a firm's capital in each category. Following the reform, high MRPK firms relatively increased the share of their capital in plants and equipment by 4 percentage points. There are no effects for low MRPK firms.

Pre-trends. To assess whether these heterogeneous effects are driven by pretrends, we produce event study graphs. We create indicator variables for being observed five years before a reform, four years before, and so on and interact these with being in a treated industry and being a high MRPK firm in a treated industry. We include the same controls as in Table 4. Figure 2 reports the relative effects by year of being a high MRPK firm in a treated industry for the logs of physical assets, MRPK, wages, and sales. The key outcomes for assessing whether misallocation declined are capital and MRPK.

Figure 2: Event Study Graphs for the Relative Effect of Foreign Capital Liberalization on High MRPK Firms



This figure reports event study graphs for the relative effects of the liberalization on firms with high pre-treatment MRPK relative to those with low pre-treatment MRPK in treated industries relative to untreated industries. The reform is normalized to take place in year 1. Each dot is the coefficient on the interaction between being observed t years after the reform and being a high MRPK firm in a treated industry. All dependent variables are in logs. The confidence intervals are at the 95% level.

Two facts are noteworthy. First, for all the outcomes, including the main outcomes of capital and MRPK, being treated by the policy did not have a strong differential effect on high MRPK firms before the policy was adopted, providing visual evidence that pre-trends are not driving the results. The lack of correlation between high MRPK firms' outcomes and the reform prior to the year of deregulation also provides some preliminary evidence that the results are not driven by mean reversion, an alternative explanation that we explore in more detail in Section 5.5.

Second, the effect of the liberalization is progressive over time, consistent with the idea that changes in the allocation of resources (such as the adjustment of worker flows and adaptation of production tools) are likely slow-moving, particularly in India (e.g. Topalova, 2010). In addition, some of the changes in allocative

Dependent Variable	TFPQ	TFPQ
	(1)	(2)
$\overline{Reform_{jt}}$	0.157 (0.166)	$0.106 \\ (0.143)$
$Reform_{jt} \times I_i^{High MRPK}$		0.084 (0.074)
Fixed Effects Firm Firm Age Size ×Year	√ √ √	√ √ √
Observations	43,791	43,791

Table 5: Effect of Foreign Capital Liberalization on TFPQ

All dependent variables are in logs. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995–2000 is above the 4-digit industry median. MRPK is estimated with the Revenue/K method. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. TFPQ is measured by estimating revenue production functions using the methodology of Levinsohn and Petrin (2003) and subtracting log average price from log TFPR. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

efficiency, might also come from competitive effects, which also happen progressively over time.

To provide further evidence in favor of our identification strategy, we also plot event study graphs separately for high and low MRPK firms for each of the four outcomes. Figure A2 reports the results. Consistent with our previous estimates, the reform has no effect on low MRPK firms across outcomes, while high MRPK firms' outcomes change sharply following the reform.

TFPQ. Motivated by the literature documenting a relationship between FDI and within-firm productivity due to technological transfers (see Keller and Yeaple (2009) and Gorodnichenko, Svejnar, and Terrell (2014)), Table 5 estimates the effects of the reform on TFPQ. While the reform changed the allocation of inputs across the firms, we cannot reject a zero effect on within-firm average productivity (column 1). Though imprecise, the point estimate is consistent with a positive effect. Similarly, when we interact the reform with the indicator variable for high MRPK (column 2), we do not find any statistically significant differential effect. However, to the extent that the effects of the policy on TFPQ are positive, when we estimate the effects of reducing misallocation on aggregate productivity in Section 6, we may be underestimating the aggregate productivity gains from the policy.

Dependent Variable	Revenues	Capital	Wages	MRPK	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_{jt}^{High MRPK} \times Financial Development_s$	-0.095 (0.078)	-0.233^{***} (0.084)	-0.189^{***} (0.060)	0.138^{***} (0.047)	
$Reform_{jt} \times I_{jt}^{High\ MRPK}$	0.203^{**} (0.079)	$\begin{array}{c} 0.527^{***} \\ (0.092) \end{array}$	$\begin{array}{c} 0.259^{***} \\ (0.064) \end{array}$	-0.349^{***} (0.121)	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	$53,\!109$	$54,\!692$	$53,\!852$	$51,\!873$	

Table 6: Heterogeneity by Local Financial Development

All dependent variables are in logs. $Reform_{jt}$ is an indicator variable equal to one if the industry has liberalized access to the international capital market. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995–2000 is above the 4-digit industry median. MRPK is calculated using the Revenue/K method. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Local financial development is proxied using the log average amount of bank credit in the state in the pretreatment period. All double and single interactions of the triple-differences specification are included in the regressions. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Importance of the local banking market. Our results so far show that opening-up to foreign capital allows high MRPK firms to invest more and grow faster. If foreign capital is acting as a substitute for a more efficient domestic banking sector, a natural implication is that firms located in areas with more developed local banking markets prior to the reform should benefit less from the reform. We directly test this hypothesis by creating a variable *Financial Developments*, defined as the log average over 1995–2000 of all bank credit in state s. We then interact this measure with all the single and cross-terms in equation (5). The variable is demeaned to restore the baseline effect on $Reform_{jt} \times I_i^{High MRPK}$. The coefficient of interest is the coefficient for the triple interaction $Reform_{jt} \times I_i^{High MRPK} \times Financial Development_s$, which captures the differential effect of the policy on high MRPK firms located in more developed local banking markets.

Table 6 reports the results.²⁹ For capital and wages, the interaction $I_i^{High MRPK} \times Reform_{jt} \times Financial Development_s$ is negative and significant at the 1% level. For MRPK, the triple interaction is positive and significant at 1%. Taken together, these results imply that capital wedges fell more following the reform for high MRPK firms located in less financially developed states.

^{29.} The sample sizes are somewhat reduced from Table 4 since state information is not available for all firms.

In addition to being statistically significant, the magnitudes of the heterogeneous effects are economically meaningful. If we focus on the change in the marginal revenue products of capital (column 4), ex-ante high MRPK firms whose state is at the 25^{th} percentile of the bank credit distribution experience a decrease in MRPK of 45% ($-0.349 + (0.138 \times -0.71)$). In contrast, high MRPK firms whose state is at the 75^{th} percentile of the bank credit distribution experience a decrease in MRPK of 16% ($-0.349 + (0.138 \times 1.37)$). Thus, the reduction at the 25^{th} percentile is roughly three times larger than the one at the 75^{th} percentile.

The fact that the effects of the policy are smaller in states where credit constraints were a priori lower further shows that opening up to foreign capital relaxed credit constraints and allowed previously constrained firms to invest more. These results also provide further evidence that our main results are not driven by differences in the relative trends between high and low MRPK firms in treated vs. untreated industries. For differential trends to explain these results, they would have to vary systematically with states' financial development. These results also suggest that under-developed domestic banking markets are an important source of misallocation in India (consistent with Krueger et al., 2002) and that foreign capital can act as a substitute.³⁰ Consistent with our interpretation that the domestic banking sector is less able to efficiently allocate capital in less financially developed states, Figure A3 shows that the average dispersion in MRPK before the reform is higher in states with lower domestic financial development.

5.4 Product Outcomes

We next estimate the effects of the policy on product-level outcomes, including prices and output. Opening-up to foreign capital can reduce prices for two reasons. If liberalization reduced the wedges on capital for high MRPK firms, these firms' marginal costs would fall. Lower marginal costs may be passed on to consumers in the form of lower prices. In addition, by allowing high MRPK firms to invest more and expand, the reform could also increase competition in the product market, leading firms to reduce their mark-ups and cut their prices.

Using product-level data on prices and output, we use the same identification strategy as before but now control for product-firm fixed effects. With these fixed

^{30.} Anne Krueger, deputy managing director of the IMF during the time of the reforms we study, wrote that in India, "banks are considered to be very high cost and inefficiently run" and that, "enabling [Indian banks] to allocate credit to the most productive users, rather than by government allocation, would make a considerable contribution to the Indian economy's growth potential" (Krueger et al., 2002).

Dependent Variable	Pri	ce	Out	put	Log(# Products) Pr(Addition) Pr(Deletion)	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
$Reform_{jt}$	-0.171***	-0.084	0.268***	0.097	0.004	-0.075***	-0.005	
	(0.031)	(0.053)	(0.058)	(0.087)	(0.025)	(0.016)	(0.035)	
$Reform_{jt} \times I_i^{High MRPK}$		-0.123*		0.242*	0.034**	0.097***	-0.051	
v <u>j</u>		(0.065)		(0.125)	(0.014)	(0.020)	(0.048)	
Fixed Effects								
Firm	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	
$Firm \times Product$	\checkmark	\checkmark	\checkmark	\checkmark	_	_	_	
Observations	$103,\!035$	$103,\!035$	$103,\!974$	$103,\!974$	32,660	32,660	$32,\!660$	

Table 7: Effect of Foreign Capital Liberalization on Product Outcomes

In columns 1-4, each observation is at the firm-product-year level. In columns 5-7, each observation is at the firm-year level. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995–2000 is above the 4-digit industry median. MRPK is calculated using the Revenue/K method. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

effects, the regressions are identified by changes in prices or output for a given product produced by a firm. Thus, the results are not biased by the addition or the deletion of products. Columns 1–2 of Table 7 report the results. On average, the reform reduces prices by 17% (column 1). Column 2 shows that the reduction is mainly driven by high MRPK firms, who reduce their prices (in total) by 21% (= -0.084 - 0.123).

We also test whether the increase in revenues caused by the reform is accompanied by a product-level increase in output. An increase in output for high MRPK firms does not need to occur mechanically in the data, since the results we have shown previously are for firm-level sales. Separately reported unit-level sales and prices are used to calculate output. Columns 3–4 of Table 7 report the effect of the reform on product-level output, which increases by 27% on average. The average effect masks considerable heterogeneity: high MRPK firms increased output by 24% relative to low MRPK firms, while we cannot reject a 0 effect on low MRPK firms' output.

In the last three columns of Table 7, we examine whether the policy affected the product portfolio of treated firms. Column 5 indicates that the number of products offered increased for high MRPK firms but not low MRPK firms. Low MRPK firms were less likely to add new products (column 6) but not more likely to delete products (column 7). High MRPK firms, on the other hand, were relatively more likely to offer new products. Altogether, these results are consistent with the initially high MRPK firms expanding into new areas, crowding out expansions by low MRPK firms.

5.5 Robustness

In this subsection, we describe a variety of robustness tests.

Industry-level variance. Table A3 uses 4-digit industry-level differences-indifferences regressions to measure the effect of the share of an industry that is treated on the log variance of MRPK (a standard measure of misallocation) in Provess (Panel A) and the Annual Survey of Industries (Panel B). Panel A shows that the policy increased industry-level capital (column 1) and reduced the dispersion of MRPK.³¹ This result is robust to the inclusion of fixed effects for the number of firms in an industry-year (to account for variance being estimated over a small number of firm-year observations). In addition, in column 4, we control for input and output tariffs³² and dereservation policies.³³ The results are again robust. In Panel B, we replicate the same regressions using the ASI, which has the advantage of including smaller establishments. We focus on industry-level measures instead of establishment-level measures for the ASI because it is not a panel at the small establishment level. The results are similar in the ASI (a representative sample of the universe of formal manufacturing establishments), suggesting that the results in Prowess are not driven by the selection of firms included in the data.

Mean reversion. In Table A4, we show that the results are robust to assigning high MRPK status using a shorter pre-treatment period (1995-1997 in Panel A and 1995-1998 in Panel B) or using only variation from the 2006 reform (Panel C). In all three cases, the years directly before the reform are not used to assign high MRPK status, so these results should be less affected by any mean reversion.

^{31.} We focus on 4-digit level industries for consistency with the ASI, where multiple changes in the industry coding system make assigning consistent 5-digit level industries impossible.

^{32.} We compute input and output tariffs from 1995-2010 — the period for which tariff data is available — following Goldberg, Khandelwal, Pavcnik, and Topalova (2010) and assume tariffs remained constant for the period 2010-2015. Input tariff measures are obtained by computing the weighted sum of the percent tariffs on each input used to produce a product based on the Indian input-output table.

^{33.} To develop our dereservation measure, we use the list of deregulated industries in ASICC from Boehm, Dhingra, and Morrow (2019) and create a crosswalk between ASICC and our definition of industry (NIC 2008) by using the ASI 2008–2009. For each establishment in the ASI, the data reports both the NIC code of the establishment and the list of all the products sold at the ASICC level. We compute a one to one mapping by assigning to each NIC the ASICC with the highest share of products sold.

In all three panels, we see that the estimates are similar to the baseline results in Table 4, though using only the 2006 reform somewhat reduces precision.

Additional time-varying shocks. We augment our baseline specification with different sets of industry or state-by-year fixed effects to capture various unobserved time-varying shocks. In Panel A of Table A5, we report the results using 2-digit industry-by-year fixed effects.³⁴ Here, the coefficients are identified by comparing firms in the same 2-digit industry and year; this accounts for any unobserved, time-varying, sector-level shocks, such as aggregate trade shocks and differences in input costs at the 2-digit industry level. We find that the coefficients of both $Reform_{jt}$ and $Reform_{jt} \times I_i^{High\;MRPK}$ are unaffected. In Panel B, we report the results when we control for 5-digit-by-year fixed effects. In this case, $Reform_{it}$ is absorbed, and we only exploit within industry-year variation. The coefficient for the interaction variable $Reform_{jt} \times I_i^{High\;MRPK}$ remains similar. In Panel C, we include state-by-year fixed effects to account for the possibility that some Indian states are more exposed to the reform due to their industrial composition and may have instituted policies affecting misallocation or were affected by shocks concurrent with the reform. Our point estimates remain quantitatively similar.

Controlling for dereservation laws. To assess whether dereservation policies could be driving our main results, we perform two tests, both reported in Table A6. In the odd columns, we exclude all 5-digit NIC industries that contained a product that was affected by a dereservation reform after 2000 (the year before our first episode of liberalization). Because this cuts our sample by more than half, in even columns, we create an indicator variable *Dereservation_{jt}* that is equal to one after industry *j* has been dereserved and control for it and its interaction with $I_i^{High MRPK}$. In both cases, the pattern of the point estimates is largely unchanged.

Controlling for trade liberalization. Our specification with industry-year fixed effects already partially accounts for potential bias from tariff reductions, since the trade liberalization occurred at the industry-level. However, it is possible that trade liberalization had a differential effect on high and low MRPK firms. To account for this, we include both the input and output tariff measures and their interaction with $I_i^{High MRPK}$ as controls in our main regression specification. Table A7 reports the results when we control for the output tariffs only (the odd columns) or both the output and input tariffs (the even columns). The effect of the foreign capital liberalization on high MRPK firms remains virtually unchanged.

^{34.} There are 23 distinct 2-digit industries.

Winsorizing outliers. We directly test the extent to which our results might be driven by outliers by winsorizing the data at the 5% level. We identify outliers either across industries or within each 2-digit industry. We report the results in Table A8 and show that the point estimates are similar to those without a measurement error correction.

Firm entry and exit. To test whether differential attrition could affect our results, we directly test whether the policy affected firm exit and entry using industry-level variation in the policy over time. If the policy had no effect on attrition, attrition should not bias our results. We identify entry in the data using the year of incorporation and use the last year in the dataset as a proxy for exit.³⁵ To estimate the average effect of the policy on exit and entry, we then create counts of the number of firms in a 5-digit industry-year cell that exited or entered. To estimate the differential effect on exit for high and low MRPK firms, we create these counts for industry-year-MRPK category cells. We cannot use the same strategy to test for differential entry, since, if a firm enters after 2000, we do not observe its MRPK during the pre-treatment period. Table A9 reports the results. There is little evidence that the policy affected entry and exit.³⁶

Spillovers. Cross-industry spillovers through input-output linkages across treated and non-treated industries could bias our estimates if they lead the policy to affect the outcomes of firms in non-liberalized industries. As in Acemoglu, Akcigit, and Kerr (2016), we separately measure the intensity of the spillover effects of liberalization through the input-output matrix on upstream and downstream industries, using entries of the Leontief inverse matrices as weights:

$$Upstream_{k,t} = \sum_{l} \left(Input \%_{l \to k}^{2000} - \mathbf{1}_{l=k} \right) \times Reform_{l,t}$$

and

$$Downstream_{k,t} = \sum_{l} \left(Output \%_{k \to l}^{2000} - \mathbf{1}_{l=k} \right) \times Reform_{l,t}$$

where k and l represents industries at the input-output table level, $\mathbf{1}_{l=k}$ is an indicator function for l = k, and the summation is over all industries, including

^{35.} True exit is not explicitly recorded in Prowess, since a firm may simply exit the panel because it decides to stop reporting its information to CMIE.

^{36.} This is not necessarily surprising since Prowess only includes large and medium-sized firms, for which exit and entry rates are likely to be relatively low. Indeed, in the average 5-digit industry, there are only 0.84 exit events a year and only 0.033 entry events. In more than 50% of industry-years, there are zero exits. In 95% of industry-years, there are zero entrances.

industry k itself. The notation $Input \%_{l \to k}$ represents the elements of the inputoutput matrix $\mathbf{A} = [a_{ij}]$, where $a_{ij} \equiv \frac{Sales_{j \to i}}{Sales_i}$ measures the total sales of inputs from industry j to industry i, as a share of the total inputs of industry i. The notation $Output \%_{k \to l}$ denotes the input-output matrix $\hat{\mathbf{A}} = [\hat{a}_{ij}]$, where $\hat{a}_{ij} \equiv \frac{Sales_{i \to j}}{Sales_i} = a_{ji} \frac{Sales_j}{Sales_i}$ measures the total sales of outputs from industry i to industry j, as a share of the total sales of industry i. We use the input-output matrices in 2000 since it is the last pre-treatment year and subtract the direct policy effects by controlling directly for the policy change in industry k in the regression.

Table A10 reports the average and heterogeneous effects, controlling for the upstream and downstream effects, and shows that they are unchanged.

Minimal controls & unfiltered data. We next show that our results are robust both to including only firm and year fixed effects as controls (removing all additional controls) and to retaining the full set of observations rather than dropping observations if firms contract their year-to-year revenues by more than 85%. Table A11 reports the results with the minimal controls, and Table A12 reports the results with the unfiltered sample. The estimates are again virtually unchanged.

Heterogeneous effects by firm risk. One potential explanation for our estimates is that firms with high ex-ante MRPK were relatively riskier for domestic lenders since they were more correlated with the Indian economy. Then, after the reform, foreign capital flowed to these firms. Indeed, our misallocation framework nests this possibility, since wedges may represent the constraints that kept foreign lenders from lending to risky firms. To test this hypothesis, in Table A13, we calculate firms' pre-treatment correlation between revenue growth and the Indian economy and control for this measure interacted with the reform. The reform did not have a differential effect on ex-ante riskier firms, and controlling for this relationship has little impact on our estimates.

Alternative cut-off. In Table A14, we report our baseline specification when we split ex-ante MRPK along the mean instead of the median (odd columns) or when we split the data into ex-ante MRPK terciles instead of using the median (even columns) and find similar results. The use of the tercile split allows to test whether the effect is monotonic with the degree of ex-ante misallocation in a non-parametric way, as the coefficient on each interaction $Reform_{jt} \times I_{jt}^{High MRPK-Tercile=x}$ gives the marginal effect for tercile x relative to the first tercile. For all outcomes, the effect shows an relative increase over each tercile, implying that firms that are "medium" constrained (second tercile) respond more than firms that are less constrained (first

Dependent Variable	Revenues Capita		Wages	MRPL	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_i^{High MRPL}$	-0.039 (0.086)	$0.198 \\ (0.119)$	$\begin{array}{c} 0.237^{***} \\ (0.069) \end{array}$	-0.276^{***} (0.061)	
$Reform_{jt}$	$\begin{array}{c} 0.120 \\ (0.105) \end{array}$	0.121^{*} (0.072)	$0.024 \\ (0.104)$	$0.096 \\ (0.096)$	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	$43,\!407$	$43,\!384$	$43,\!407$	$43,\!407$	

Table 8: Effect of Foreign Capital Liberalization by Firms' Ex-ante MRPL

All dependent variables are in logs. High MRPL firms are defined in an analogous way to high MRPK firms using the Revenue/L method. $Reform_{jt}$ is an indicator variable equal to one if the industry has liberalized access to the foreign capital market. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

tercile) but not as much as firms that are "very" constrained (third tercile).

5.6 Extension to Labor Misallocation

Our results so far show that opening up to foreign capital allowed firms not only to invest more (as seen by the increase in their capital stock) but also to expand their wage bills. Reducing capital market frictions may simply increase the demand for labor because of the complementarity between capital and labor in the production function. However, it is also possible that the financial liberalization directly reduced labor misallocation, a hypothesis that we test in this section.

Although labor is often modelled as a fully adjustable variable input across periods, in reality, labor is likely to have a fixed-cost component due to wage rigidity and hiring/firing costs. As a result, when there is a mismatch between the payments to labor and the generation of cash-flows, financial constraints may affect employment and labor (mis)allocation. Schoefer (2015) and Fonseca and Doornik (2021) provide evidence in support of this channel.

To investigate if the reform reduces labor misallocation, we use the same estimation strategy as before but now compare the effects of the policy on firms with higher or lower marginal revenue products of labor (MRPL) prior to the reform. We classify high and low MRPL firms analogously to how we classify high and low MRPK firms and estimate the heterogeneous effects of the reform on high MRPL firms. Table 8 reports the results. Following the reform, high MRPL firms relatively increase their total wage bill (column 3) by 24%. Among ex-ante high MRPL firms, MRPL decreases by 28% relative to low MRPL firms (column 4). By allowing high MRPL firms to grow faster and to expand employment, the deregulation appears to have reduced labor misallocation.

6 Aggregate Effects

While our reduced-form estimates show that misallocation fell, they do not tell us whether this had economically meaningful effects on output growth. To measure the policies' aggregate effects, we now estimate bounds for the effect of the reduction in misallocation on treated industries' Solow residual, a proxy for aggregate productivity, using equation (3). Equation (3) is re-stated below:

$$\Delta Solow_{I,t} \approx \sum_{i \in I} \lambda_i \Delta \log A_i + \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_i \alpha_i^x \frac{\tau_i^x}{1 + \tau_i^x} \Delta \log x_i$$

where λ_i is the ratio of firm *i*'s sales to treated industry *I*'s net output, $\Delta \log A_i$ is the change in TFPQ, α_i^x is the output elasticity with respect to x, τ_i^x is the level of firm-specific input wedges prior to the policy change, and $\Delta \log x_i$ is the change in the log input x consumed by firm i.

6.1 Identification

Equation (3) shows that the Solow residual can increase for two reasons: (1) individual firms become more productive (within-firm productivity) or (2) inputs either increase for producers with positive wedges or decrease for producers with negative wedges (firm-level inputs). We discuss each part in turn.

Within-firm productivity. The contribution of the change in within-firm productivity to the Solow residual is given by $\sum_{i \in I} \lambda_i \Delta \log A_i$. Since we do not observe a significant effect of the policy on TFPQ in the difference-in-differences regressions (see Table 5), we set $\Delta \log A_i = 0$. **Firm-level inputs.** The contribution of changing firm-level inputs to the Solow residual is given by:

$$\Delta Solow_{I,t} = \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_i \alpha_i^x \frac{\tau_i^x}{1 + \tau_i^x} \,\Delta \log x_i. \tag{6}$$

Note that an increase in inputs for some firms does not need to mechanically increase the Solow residual. The wedge τ_i^x can be negative for firms whose capital is subsidized, zero if there is no distortion, or positive for firms that face distortions in accessing capital. An increase in inputs if there is no misallocation will have no effect on the Solow residual, since, in that case, τ_i would be zero for all firms. Similarly, if the policy increases misallocation by increasing inputs x_i for firms with negative wedges, the contribution to the Solow residual would be negative even though inputs increase.

To estimate (6), observe that most components of this expression are readily identifiable in the data or given by our estimates from the natural experiment. The sales shares of net output λ_i can be computed using input-output data,³⁷ and α_i^x can be estimated using the production function estimation. Under the standard difference-in-differences assumption that untreated industries are unaffected by the policy, $\Delta \log x_i$ can be predicted from difference-in-differences regressions with heterogeneous effects where log usage of each input is the outcome variable.³⁸

We next turn to identifying τ_i^x . Equation (6) highlights that errors in the estimation of τ_i^x , the level of firm-specific input wedges prior to the policy change, can bias the aggregate policy effects, as $\frac{\tau_i^x}{1+\tau_i^x}$ is multiplicative with $\Delta \log x_i$ and increasing in τ_i^x . If we use cross-sectional variation prior to the policy change to identify τ_i^x , measurement error could lead to greater dispersion in these values. Since we have shown in Section 5 that the reform has a positive effect on inputs for firms with relatively greater wedges, inflated values of $\frac{\tau_i^x}{1+\tau_i^x}$ would be multiplied by the positive predicted change in inputs.

To circumvent this potential bias, we generate lower and upper bound estimates of the policies' effect. We first note that if the policy strictly reduces misallocation, then the aggregate effects are strictly increasing in τ_i^x (as discussed above). Then, a

^{37.} To measure total sales by sector I not re-used by firms in I as inputs, we sum over treated firms' total sales in 2000 (the last pre-treatment year). We then use information from the Annual Survey of Industries to compute the share of output that is re-used by the treated industries as inputs and scale total sales by 1 minus this value. Finally, λ_i is calculated for a firm i by dividing a firm i's sales by this value.

^{38.} The difference-in-differences assumption could be partially relaxed by modeling spillovers explicitly and estimating spillovers effects.
lower bound for τ_i^x generates a lower bound aggregate effect, while an upper bound for τ_i^x generates an upper bound aggregate effect. To generate an upper bound, we can attribute *all* observed pre-treatment deviations from efficiency to wedges. As we will see, under the assumption that the policy strictly reduced misallocation, we can also generate a lower bound value for τ_i^x . While this assumption may not always be reasonable, our reduced-form empirical results, which show that the policy causally reduced MRPK and MRPL for firms that had ex-ante above median values of MRPK and MRPL, provide strong evidence in its favor.

Identifying & estimating the lower bound of τ_i^x . By definition, the postpolicy wedge for a firm is always given by: $\tau_{post}^x = \tau_{pre}^x + \Delta \tau^x$, where $\Delta \tau^x$ is the change in τ^x due to the policy and *i* is suppressed to simplify notation. Under the assumption of no increase in misallocation after the policy (i.e. the policy does not subsidize firms), $\tau_{post}^x \ge 0$, which implies that $\min_{\tau_{post}^x \ge 0} \tau_{pre}^x = -\Delta \tau^x$. Thus, the minimum possible pre-treatment wedge is given by the scenario where, after the policy change, the industry is efficient, and there are no wedges left. In this case, any measured dispersion in marginal revenue products after the policy change is attributed to mismeasurement and misspecification as opposed to misallocation. So estimating $\Delta \tau^x$ gives us a lower bound estimate of τ_{pre}^x , and we can apply equation (6) to estimate a lower bound of the first order effects of the policy on the Solow residual.

Since the minimum values of the pre-reform wedges τ^x are given by the change in the wedges due to the policy, and since in our formula, wedges vary at the firm level, we can predict the minimum firm-specific wedges with a difference-indifferences regression with heterogeneous effects where the outcome variable is the marginal revenue product of input x. For example, in the case of τ_i^k , we estimate

$$\log MRPK_{ijt} = g(Reform_{jt}, \mathbf{C}_{it}) + \Gamma \mathbf{X}_{it} + \alpha_i + \delta_t + \epsilon_{ijt}$$
(7)

where $g(Reform_{jt}, \mathbf{C}_{it})$ is a flexible function of $Reform_{jt}$ and firm characteristics \mathbf{C}_{it} . Since we focus on within-industry changes in allocation, allowing the effect of $Reform_{jt}$ to depend on firm characteristics is important, as it allows our estimates of τ_i^k to vary within an industry j. As shown in Appendix B, if the policy completely eliminated misallocation, $\hat{g}(1, \mathbf{C}_{it})$ is an unbiased predictor of $\log(1 + \tau_i^k)$. Then, τ_i^k can be estimated by computing $\hat{\tau}_i^k = e^{\hat{g}_i(1, \mathbf{C}_{it})} - 1$. An analogous process can be used to estimate the wedges on labor. Appendix C discusses the settings where this methodology can be applied to estimate lower bound aggregate effects

with natural experiments.

As discussed in Section 4.3, estimating the change in wedges using a differencein-differences specification is less sensitive to the issues that occur when crosssectional data are used to estimate distortions. To the extent that firms' measurement error is time-invariant over the period of our experiment, it will absorbed by the firm fixed effects α_i . Time-varying macro-economic shocks or economy-wide changes in markups or the costs of inputs will be absorbed by year fixed effects, as well as year fixed effects interacted with firm characteristics. Additionally, the effects of time-varying shocks to marginal revenue products, such as productivity shocks, even if they are not economy-wide, will *not* be attributed to the reform, as long as the standard difference-in-differences assumption holds and they are uncorrelated with *Reform*_{it} conditional on the controls.

In practice, since we observe larger effects on inputs and marginal revenue products for firms with ex-ante higher marginal revenue products, we specify $g(Reform_{jt}, \mathbf{C}_{it})$ to allow for heterogeneous effects for firms with above median pre-treatment values of MRPK and MRPL.³⁹ We use analogous regression specifications to estimate the change in inputs due to the policy. Table A15 reports the results of the regressions used to identify both the change in wedges and the change of inputs. Following the identifying assumption in the production function estimation used to identify TFPQ, we assume that materials are not misallocated $(\tau_i^m = 0 \text{ for all } i).^{40}$

Identifying & estimating the upper bound of τ_i^x . For our upper bound measure of τ_i^x , we attribute all the pre-treatment, cross-sectional observed deviations from efficiency in the data to misallocation. While we observe pre-treatment measures of MRPK directly, $MRPK = (1 + \tau_i^x)p_i^x$. Since p_i^x is not observable, observing MRPK alone does not directly identify the level of the wedges. Instead,

39. For example, for the marginal revenue product of capital, we estimate

$$\begin{split} \log MRPK_{ijt} = & \beta_1 Reform_{jt} + \beta_2 Reform_{jt} \times I_i^{High \ MRPK} + \beta_3 Reform_{jt} \times I_i^{High \ MRPL} \\ & + \mathbf{\Gamma} \mathbf{X}_{it} + \alpha_i + \delta_t + \epsilon_{ijt}. \end{split}$$

We can then predict $\hat{\tau}_i^k$ by computing:

$$\widehat{\log(1+\tau_i^k)} = \hat{\beta}_1 \operatorname{Reform}_j + \hat{\beta}_2 \operatorname{Reform}_j \times I_i^{\operatorname{High} MRPK} + \hat{\beta}_3 \operatorname{Reform}_j \times I_i^{\operatorname{High} MRPL},$$

where $Reform_j$ is an indicator variable equal to 1 if a firm is in an industry that liberalized between 1995 and 2015.

40. In practice, relaxing this assumption and calculating wedges and changes in inputs for materials the same way we do for capital and labor has a small but positive effect on the estimated change in the Solow residual.

to identify the levels of the wedges, we use the relationships $\tau_i^K = \alpha_i^K \frac{p_i y_i}{rK_i} - 1$ and $\tau_i^L = \alpha_i^L \frac{p_i y_i}{wL_i} - 1$, where r is the rental rate of capital and wL_i is the wage bill.⁴¹ The wage bill wL_i and sales $p_i y_i$ are observable in the last pre-treatment year (2000) in Prowess, and as before, α_i^K and α_i^L are given by production function estimation. For the rental rate of capital r, we follow Hsieh and Klenow (2009) and set r = 10%. In practice, since the capital wedges are decreasing in r, the estimated aggregate effect will also be decreasing in r. Thus, our choice of a relatively low value of r = 10% for India is consistent with our goal of calculating an upper bound.

6.2 Results

Bounds on the first order approximation. Having estimated all the components of equation (3), we calculate that the lower bound increase in the treated industries' Solow residual is 3.4% (see row 1 of Table 9), and the upper bound is 16.3% (row 2).

Comparison with alternative estimates. For robustness, we next compare this lower bound estimate to several other estimates of the aggregate effect of the policy. In row 3, we combine the CES model of Hsieh and Klenow (2009), which translates changes in the log variance of TFPR into changes in aggregate TFP, with our reduced-form estimates.⁴² This allows us to obtain a direct estimate of the aggregate effects of the policy rather than a first order approximation, at the cost of stronger assumptions about firm's production functions, demand, and the distribution of wedges. This model indicates that treated industries' aggregate TFP increased by 8.6%.

We next consider the cumulative effect of the reforms. The estimates of the reforms' effects over time in Figure 2 suggest that the effects on inputs and the wedges grew over time. Thus, using estimates from a standard difference-indifferences that assumes constant treatment effects over time may lead row 1 of Table 9 to underestimate the long-run effects of the policies. Since the effects

^{41.} These relationships come from firms' cost-minimization problems. A cost-minimizing firm sets $\frac{rK(1+\tau_i^K)}{p_i y_i} = \alpha_i^K$ and $\frac{wL_i(1+\tau_i^L)}{p_i y_i} = \alpha_i^L$. An efficient firm sets $\frac{rK}{p_i y_i} = \alpha_i^K$ and $\frac{wL_i}{p_i y_i} = \alpha_i^L$, so the wedges are the taxes or subsidies that would make firms' decisions appear efficient.

^{42.} In the model of Hsieh and Klenow (2009), $\log TFP = -\sigma/2Var(\log TFPR_i)$, where σ is assumed to be 3, and $\log(TFPR_i) = \alpha_i^K \log MRPK_i + \alpha_i^L \log MRPL_i$. We can use our estimates of $\log MRPK_i$ and $\log MRPL_i$ in 2000 to directly predict $\log TFPR_i$ and then estimate its variance. To get the post-treatment $\log TFPR_i$, we can simply use the regression estimates in Table A15 to predict the firm-level change in $\log MRPK_i$ and $\log MRPL_i$ and add the predicted $\Delta \log TFPR_i$ to the pre-treatment value of $\log TFPR_i$.

	Increase in Solow Residual
Lower Bound	3.4%
Upper Bound	16.3%
Hsieh-Klenow Model	8.6%
Lower Bound Allowing for Cumulative Effects	6.2%
Non-Linear Approximation	6.0%

 Table 9: Effects of Foreign Capital Market Liberalization on the Solow Residual

 of Treated Industries

This table reports the estimates of the effect of the foreign capital liberalizations in 2001 and 2006 on treated industries' Solow residual. Rows 1, 2, and 4 use a first order approximation (equation (3)). Row 3 uses a model in the style of Hsieh and Klenow (2009), and the last row uses a non-linear approximation, described in Appendix D. The estimates are generated using the Prowess dataset. Rows 1, 4, and 5 identify the wedges by assuming the policy eliminated misallocation. Rows 2 and 3 use cross-sectional data to identify the baseline wedges.

plateau after 5 years in Figure 2, we re-calculate the lower bound approximation using the estimated policy effect five years after the reforms. This yields a larger estimate (6.2%, row 4).

Since the first order approximation may not be a good approximation if there are important higher order effects of the policy on the Solow residual, we also construct a non-linear approximation of the policies' effects on the Solow residual by estimating policy effects year-by-year and chaining the results. Appendix C describes this process. Since the non-linear approximation requires estimating dynamic policy effects over five years, it should be compared to the cumulative estimate in row 4. The non-linear approximation (6.0%, reported in row 5) is quite close to the simpler, cumulative first order approximation.

Discussion of magnitudes. Our ability to benchmark the size of our estimates is limited by the sparsity of the literature on misallocation and foreign capital liberalization. However, Bollard, Klenow, and Sharma (2013) and Sivadasan (2009) do both estimate the effects of the 1991 FDI liberalizations. While these liberalizations occurred during a macroeconomic crisis, complicating the interpretation of their estimates, both papers find at least some evidence that FDI liberalization led to large increases in aggregate productivity in affected sectors (on the order of or even larger than the range of potential aggregate effects we report). While both papers attribute the majority of these gains to within-firm productivity growth rather than reduced misallocation, their decompositions may systematically underestimate reallocation's contribution to productivity growth (Nishida, Petrin, Rotemberg, and White, 2017). More broadly, Topalova (2005) finds that FDI openness in India has a large negative association with district-level poverty. Thus, our estimates of the aggregate productivity effects of FDI liberalization appear to be in line with the existing literature.

7 Conclusion

Exploiting within-country, within-industry, and cross-time variation, we show that foreign capital liberalization reduced the misallocation of capital and labor in India. In doing so, this paper addresses two of the key challenges faced by the misallocation literature. First, it provides direct evidence that policymakers can change allocative efficiency and aggregate productivity. Second, it develops new tools that can be combined with estimates from natural experiments to measure the aggregate effects of policies.

The liberalization, which allowed for the automatic approval of foreign investments and raised caps on foreign equity in the 2000's, increased capital in treated industries. However, the effect of the liberalization on the average firm masks important heterogeneity. The entirety of the liberalization's effect on firms' outcomes is driven by increased investment in firms that previously had high MRPK (high sales to capital ratios). These firms' MRPK fell, indicating that the policy reduced misallocation. Thus, foreign capital liberalization can be an important tool for low-income countries to reduce capital market frictions.

Finally, variation from a natural experiment also allows us to estimate aggregate effects of reducing misallocation that – unlike cross-country or time series comparisons – are less sensitive to measurement and model misspecification error. Aggregating our reduced-form estimates, we find that the policy had economically meaningful effects, increasing treated industries' Solow residual by between 3% and 16%.

References

- Acemoglu, Daron, Ufuk Akcigit, and William Kerr. 2016. "Networks and the macroeconomy: An empirical exploration." NBER Macroeconomics Annual 30 (1): 273–335.
- Adamopoulos, Tasso, Loren Brandt, Jessica Leight, and Diego Restuccia. 2017. "Misallocation, Selection and Productivity: A Quantitative Analysis with Panel Data from China." NBER Working Paper, no. 23039.
- Adamopoulos, Tasso, and Diego Restuccia. 2014. "The Size Distribution of Farms and International Productivity Differences." American Economic Review 104 (6): 1667–1697.

- Aghion, Philippe, Robin Burgess, Stephen J Redding, and Fabrizio Zilibotti. 2008. "The unequal effects of liberalization: Evidence from dismantling the License Raj in India." American Economic Review 98 (4): 1397–1412.
- Alfaro, Laura, Areendam Chanda, Sebnem Kalemli-Ozcan, and Selin Sayek. 2004. "FDI and economic growth: the role of local financial markets." *Journal of International Economics* 64 (1): 89–112.
- Allcott, Hunt, Allan Collard-Wexler, and Stephen D O Connell. 2016. "How Do Electricity Shortages Affect Industry ? Evidence From India." American Economic Review 106 (3): 587–624.
- Amirapu, Amrit, and Michael Gechter. 2019. "Labor Regulations and the Cost of Corruption: Evidence from the Indian Firm Size Distribution." *Review of Economics and Statistics*, 1–48.
- Arce, Óscar, Sergio Mayordomo, and Ricardo Gimeno. 2020. "Making Room for the Needy: The Credit-Reallocation Effects of the ECB's Corporate QE." *Review of Finance.*
- Asker, John, Allan Collard-Wexler, and Jan De Loecker. 2014. "Dynamic Inputs and Resource (Mis)Allocation." Journal of Political Economy 122 (5): 1013–1063.
- Bai, John, Daniel Carvalho, and Gordon M. Phillips. 2018. "The Impact of Bank Credit on Labor Reallocation and Aggregate Industry Productivity." Journal of Finance 73 (6): 2787–2836.
- Banerjee, Abhijit V, and Benjamin Moll. 2010. "Why Does Misallocation Persist?" American Economic Journal: Macroeconomics 2 (1): 189–206.
- Banerjee, Abhijit V., Esther Duflo, Rachel Glennerster, and Cynthia Kinnan. 2015. "The Miracle of Microfinance? Evidence from a Randomized Evaluation." American Economic Journal: Applied Economics 7 (1): 22–53.
- Baqaee, David, and Emmanuel Farhi. 2019. "A Short Note on Aggregating Productivity." NBER Working Paper, no. 25688.
 - —. 2020. "Productivity and misallocation in general equilibrium." *Quarterly Journal of Economics* 135 (1): 105–163.
- Bartelsman, Eric, John Haltiwanger, and Stefano Scarpetta. 2013. "Cross-Country Differences in Productivity: The Role of Allocation and Selection." *American Economic Review* 103 (1): 305–334.
- Beena, PL, Laveesh Bhandari, Sumon Bhaumik, Subir Gokarn, and Anjali Tandon. 2004. "Foreign direct investment in India." In *Investment Strategies in Emerging Markets*, edited by Saul Estrin and Klaus Meyer. Edward Elgar.
- Bils, Mark, Peter Klenow, and Cian Ruane. 2018. "Misallocation or Mismeasurement?," Working paper.
- Bloom, Nicholas, Benn Eifert, Aprajit Mahajan, David McKenzie, and John Roberts. 2013. "Does Management Matter? Evidence from India." *Quarterly Journal of Economics* 128 (1): 1–51.
- Boehm, Johannes, Swati Dhingra, and John Morrow. 2019. "The Comparative Advantage of Firms." Working Paper.
- Bollard, Albert, Peter J Klenow, and Gunjan Sharma. 2013. "India's mysterious manufacturing miracle." Review of Economic Dynamics 16 (1): 59–85.

- Buera, Francisco J, Joseph P Kaboski, and Yongseok Shin. 2011. "Finance and Development: A Tale of Two Sectors." American Economic Review 101 (5): 1964–2002.
- Burbidge, John, Lonnie Magee, and Leslie Robb. 1988. "Alternative Transformations to Handle Extreme Values of the Dependent Variable." Journal of the American Statistical Association 83 (401): 123–127.
- Burgess, Robin, and Rohini Pande. 2005. "Do rural banks matter? Evidence from the Indian social banking experiment." American Economic Review 95 (3): 780–795.
- Chari, A V. 2011. "Identifying the Aggregate Productivity Effects of Entry and Size Restrictions: An Empirical Analysis of License Reform in India." *American Economic Journal: Economic Policy* 3 (2): 66–96.
- Cole, Shawn. 2009. "Financial Development, Bank Ownership, and Growth: or, Does Quantity Imply Quality?" Review of Economics and Statistics 91 (1): 33–51.
- David, Joel M, Hugo A Hopenhayn, and Venky Venkateswaran. 2016. "Information, Misallocation, and Aggregate Productivity." Quarterly Journal of Economics 131 (2): 943–1005.
- David, Joel M, and Venky Venkateswaran. 2019. "The Sources of Capital Misallocation." American Economic Review 109 (7): 2531–2567.
- Delatte, Anne Laure, Adrien Matray, and Noémie Pinardon Touati. 2019. "Private Credit Under Political Influence: Evidence from France." Working Paper, 1–40.
- Detragiache, E, T Tressel, and P Gupta. 2008. "Foreign banks in poor countries: theory and evidence." Journal of Finance 63 (5): 2123–2160.
- Dinc, I Serdar, and Nandini Gupta. 2011. "The Decision to Privatize: Finance and Politics." Journal of Finance 66 (1): 241–269.
- Duranton, D., E. Ghani, A. Goswani, and W. Kerr. 2017. "Misallocation in India." Working paper.
- Fonseca, Julia, and Bernardus Van Doornik. 2021. "Financial Development, Labor Markets, and Aggregate Productivity: Evidence from Brazil." *Journal of Financial Economics* (forthcoming).
- Garcia-Santana, Manuel, and Josep Pijoan-Mas. 2014. "The reservation laws in India and the misallocation of production factors." Journal of Monetary Economics 66:193–209.
- Goldberg, Pinelopi Koujianou, Amit Kumar Khandelwal, Nina Pavcnik, and Petia Topalova. 2010. "Imported Intermediate Inputs and Domestic Product Growth: Evidence from India." *Quarterly Journal of Economics* 125 (4): 1727–1767.
- Gollin, Douglas, and Christopher Udry. 2021. "Heterogeneity, Measurement Error, and Misallocation: Evidence from African Agriculture." Journal of Political Economy 129 (1): 1– 80.
- Gopinath, Gita, Şebnem Kalemli-Özcan, Loukas Karabarbounis, and Carolina Villegas-Sanchez. 2017. "Capital allocation and productivity in South Europe." Quarterly Journal of Economics 132 (4): 1915–1967.
- Gorodnichenko, Yuriy, Jan Svejnar, and Katherine Terrell. 2014. "When does FDI have positive spillovers? Evidence from 17 transition market economies." Journal of Comparative Economics 42 (4): 954–969.

- Guner, Nezih, Gustavo Ventura, and Yi Xu. 2008. "Macroeconomic implications of size-dependent policies." Review of Economic Dynamics 11 (4): 721–744.
- Haltiwanger, John, Robert Kulick, and Chad Syverson. 2018. "Misallocation Measures: The Distortion That Ate the Residual." NBER Working Paper, no. 24199.
- Hombert, Johan, and Adrien Matray. 2016. "The Real Effects of Lending Relationships on Innovative Firms and Inventor Mobility." *Review of Financial Studies* 30 (7): 2413–2445.
- Hsieh, Chang-Tai, and Peter J Klenow. 2009. "Misallocation and manufacturing TFP in China and India." *Quarterly Journal of Economics* 124 (4): 1403–1448.
- Kaboski, Joseph P, and Robert M Townsend. 2011. "A Structural Evaluation of a Large-Scale Quasi-Experimental Microfinance Initiative." *Econometrica* 79 (5): 1357–1406.
- Kehrig, Matthias, and Nicolas Vincent. 2019. "Good Dispersion, Bad Dispersion." NBER Working Paper, no. 25923.
- Keller, Wolfgang, and Stephen R Yeaple. 2009. "Multinational Enterprises, International Trade, and Productivity Growth: Firm Level Evidence from the United States." *Review of Economics and Statistics* 91 (4): 821–831.
- Krueger, Anne O, et al. 2002. *Economic policy reforms and the Indian economy*. University of Chicago Press.
- LaFave, Daniel, and Duncan Thomas. 2016. "Farms, Families, and Markets: New Evidence on Completeness of Markets in Agricultural Settings." *Econometrica* 84 (5): 1917–1960.
- Levinsohn, James, and Amil Petrin. 2003. "Estimating production functions using inputs to control for unobservables." *Review of Economic Studies* 70 (2): 317–341.
- MacKinnon, James G, and Magee Lonnie. 1990. "Transforming the Dependent Variable in Regression Models." International Economic Review 31 (2): 315–39.
- Martin, Leslie A., Shanthi Nataraj, and Ann E. Harrison. 2017. "In with the big, out with the small: Removing small- scale reservations in India." *American Economic Review* 107 (2): 354–386.
- Midrigan, Virgiliu, and Daniel Yi Xu. 2014. "Finance and Misallocation: Evidence from Plant-Level Data." American Economic Review 104 (2): 422–458.
- Moll, Benjamin. 2014. "Productivity Losses from Financial Frictions: Can Self-Financing Undo Capital Misallocation?" American Economic Review 104 (10): 3186–3221.
- Nishida, Mitsukuni, Amil Petrin, Martin Rotemberg, and T. Kirk White. 2017. "Are We Undercounting Reallocation's Contribution to Growth?" Working Paper.
- Panagariya, Arvind. 2008. India: The emerging giant. Oxford University Press.
- Petrin, Amil, and James Levinsohn. 2012. "Measuring aggregate productivity growth using plantlevel data." The RAND Journal of Economics 43 (4): 705–725.
- Restuccia, Diego, and Richard Rogerson. 2008. "Policy distortions and aggregate productivity with heterogeneous establishments." *Review of Economic Dynamics* 11 (4): 707–720.
- 2017. "The Causes and Costs of Misallocation." Journal of Economic Perspectives 31 (3): 151–174.

- Restuccia, Diego, Dennis Tao Yang, and Xiaodong Zhu. 2008. "Agriculture and aggregate productivity: A quantitative cross-country analysis." *Journal of Monetary Economics* 55 (2): 234–250.
- Rotemberg, Martin. 2019. "Equilibrium effects of firm subsidies." American Economic Review 109 (10): 3475–3513.
- Rotemberg, Martin, and Kirk White. 2017. "Measuring Cross-Country Differences in Misallocation," Working paper.
- Schoefer, Benjamin. 2015. "The financial channel of wage rigidity," Working paper.
- Singh, Kulwindar. 2005. "Foreign direct investment in India: A critical Analysis of FDI from 1991-2005." Centre for Civil Society, New Delhi.
- Sivadasan, Jagadeesh. 2009. "Barriers to competition and productivity: Evidence from India." BE Journal of Economic Analysis & Policy 9 (1).
- Sraer, David Alexandre, and David Thesmar. 2020. "A Sufficient Statistics Approach for Aggregating Firm-Level Experiments." NBER Working Paper.
- Topalova, Petia. 2005. "Trade Liberalization, Poverty, and Inequality: Evidence from Indian Districts." BNER Working Paper.
 - —. 2007. "Trade liberalization, poverty and inequality: Evidence from Indian districts." In Globalization and poverty, 291–336. University of Chicago Press.
- ———. 2010. "Factor immobility and regional impacts of trade liberalization: Evidence on poverty from India." *American Economic Journal: Applied Economics* 2 (4): 1–41.
- Topalova, Petia, and Amit Khandelwal. 2011. "Trade Liberalization and Firm Productivity: The Case of India." *Review of Economics and Statistics* 93 (3): 995–1009.
- Townsend, Robert. 1994. "Risk and Insurance in Village India." Econometrica 62:539-591.
- Udry, Christopher. 1996. "Gender, Agricultural Production, and the Theory of the Household." Journal of Political Economy 104 (5): 1010–1046.
- Varela, Liliana. 2017. "Reallocation, Competition, and Productivity: Evidence from a Financial Liberalization Episode." *Review of Economic Studies* 85 (2): 1279–1313.
- Wooldridge, Jeffrey M. 2009. "On estimating firm-level production functions using proxy variables to control for unobservables." *Economics Letters* 104 (3): 112–114.
- Xu, Chenzi. 2020. "Reshaping Global Trade: The Immediate and Long-Run Effects of Bank Failures." *Working Paper.*

Online Appendix

Appendix A: Derivation of Aggregation Formula

In this section, we derive equation (3), the formula used to approximate the change in the Solow residual due to the policy. We start by defining

 $y_i = A_i f_i \left(\{ y_{ij} \}_j \right),$

where y_i is the output of firm i, A_i is firm i's productivity, f_i is the production function, and y_{ij} is the amount of input j used by firm i. Then, the total derivative of y_i is

$$d\log y_i = \sum_j \frac{\partial \log f_i}{\partial \log y_{ij}} d\log y_{ij} + d\log A_i.$$
(8)

A firm i solves the cost minimization problem

$$\mathcal{C}_i(p, y_i) = \sum_j p_j y_{ij} + \gamma_i (y_i - A_i f_i \left(\{ y_{ij} \}_j \right), \tag{9}$$

where p is the vector of prices, p_j is the price of a good produced by j, and γ_i is the Lagrange multiplier. From the first order conditions of equation (9)

$$p_j = \gamma_i A_i \frac{\partial f_i}{\partial y_{ij}}.$$
(10)

Then,

$$\mu_i = \frac{p_i}{\partial \mathcal{C} / \partial y_i} = \frac{p_i}{\gamma_i},$$

where μ_i is the output wedge of *i* (price over marginal cost), implying that $\gamma_i = \frac{p_i}{\mu_i}$. Substituting this relationship into equation (10) shows that $p_j = \frac{p_i}{\mu_i} A_i \frac{\partial f_i}{\partial y_{ij}}$. Then

$$\frac{p_j y_{ij}}{p_i y_i} = \frac{A_i y_{ij}}{\mu_i y_i} \frac{\partial f_i}{\partial y_{ij}} \\ = \frac{\partial \log f_i}{\partial \log y_{ij}} \frac{1}{\mu_i},$$

which can be rewritten as $\mu_i \frac{p_j y_{ij}}{p_i y_i} = \frac{\partial \log f_i}{\partial \log y_{ij}}$. Then, substituting this into the total derivative (equation (8)) produces

$$d\log y_i = d\log A_i + \mu_i \sum_j \frac{p_j y_{ij}}{p_i y_i} d\log y_i.$$

Note that this implies that

$$\frac{1}{\mu_i} (d\log y_i - d\log A_i) - \sum_{j \notin I} \frac{p_j y_{ij}}{p_i y_i} d\log y_{ij} = \sum_{j \in I} \frac{p_j y_{ij}}{p_i y_i} d\log y_{ij}.$$
 (11)

Now that we have these expressions, we can turn to deriving our object of interest. We define firm-level net output to be c_i and total nominal industry-level output to be $PC = \sum_{i \in I} p_i c_i$, where $c_i = y_i - \sum_{j \in I} y_{ji}$. Then

$$d\log c_i = \frac{y_i}{c_i} d\log y_i - \sum_{j \in I} \frac{y_{ji}}{c_i} d\log y_{ji}.$$

The change in industry-level net output is defined by

$$d\log C = \sum_{i} \frac{p_i c_i}{PC} d\log c_i,$$

where after substitution, we get

$$d\log C = \sum_{i} \frac{p_i c_i}{PC} d\log c_i = \sum_{i} \left(\frac{p_i y_i}{PC} d\log y_i - \sum_{j \in I} \frac{p_i y_{ji}}{PC} d\log y_{ji} \right).$$

Then, the change in the Solow residual for I is given by

$$dSolow_I = d\log C - \sum_{i \in I} \sum_{j \notin I} \frac{p_j y_{ij}}{p_i y_i} \frac{p_i y_i}{PC} d\log y_{ij}.$$

Using equation (11), with a little algebra, we can rewrite this as

$$dSolow_I = \sum_{i \in I} \lambda_i (1 - \frac{1}{\mu_i}) (d\log y_i - d\log A_i) + \sum_{i \in I} \lambda_i d\log A_i,$$
(12)

where $\lambda_i = \frac{p_i y_i}{PC}$. Now, we transform equation (12) to use input wedges instead of output wedges, so that it matches equation (3). This allows us to rewrite equation (12) in terms of firm-level capital, labor, and materials wedges where each firm-

input combination is a "producer."⁴³ The wedge on firm *i*'s input *x* is τ_i^x , and the price paid by the firm is $(1 + \tau_i^x)p^x$, while the marginal cost of producing *x* is p^x . The gross output wedge for producer (x, i) is given by: $\mu_i^x = 1 + \tau_i^x$. Then, for a given firm *i*, $d \log y_i - d \log A_i = \sum_{x \in \{K, L, M\}} \alpha_i^x d \log x_i$, where α_i^x is the output elasticity with respect to input *x*. So, we can rewrite equation (12) as:

$$dSolow_{I,t} = \sum_{i \in I} \lambda_i d \log A_i + \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_i \alpha_i^x \left(1 - \frac{1}{1 + \tau_i^x}\right) d \log x_i,$$

which in turn simplifies to

$$dSolow_{I,t} = \sum_{i \in I} \lambda_i d \log A_i + \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_i \alpha_i^x \frac{\tau_i^x}{1 + \tau_i^x} d \log x_i.$$

To implement the first order approximation, for any variable x, we use discrete changes Δx instead of infinitesimal changes dx. Then the first order approximation is given by

$$\Delta Solow_{I,t} \approx \sum_{i \in I} \lambda_i \Delta \log A_i + \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_i \alpha_i^x \frac{\tau_i^x}{1 + \tau_i^x} \Delta \log x_i.$$

^{43.} While equation (12) models wedges on output rather than inputs, this framework is general and input wedges can be thought of as a special case of this formulation. In particular, we can think of each input wedge for firm i coming from a fictitious middleman firm that buys the input without a wedge and then sells it with an output wedge to firm i.

Appendix B: Estimating the Distribution of the Minimum Wedges

In this appendix, we show that the difference-in-differences regressions with heterogeneous effects can be used to estimate the minimum wedge prior to the policy under the assumptions outlined in the main text. We focus here on estimating τ_{pre}^k , where the *i* subscript is surpressed for notational simplicity. The reasoning is identical for labor and materials.

Denote $mrpk_i$ the true marginal revenue product of capital of firm i (which is never observed) and $MRPK_i$ the marginal revenue product of capital observed in the data with measurement error, such that we have:

$$\log(MRPK_{it}) = \log(mrpk_{it}) + \mu_i + \eta_t + \epsilon_{it}$$

where ϵ_{it} is a firm-period idiosyncratic error, μ_i is a firm-specific, time-invariant shock, and η_t is a time-period specific shock.

Denote T_j to be the time period of the reform in a disaggregated industry j. If a firm is in an industry that does not go through a reform $(Reform_{jt} = 0 \forall t)$ or if the firm is in an industry that will be reformed but the reform has not taken place yet $(t < T_j)$:

$$\log(mrpk_{ijt}) = \log(1 + \tau_{it}^k) + \log(p_t^k).$$

Under the assumption the policy has eliminated misallocation, if the firm is in an industry that is reformed and the reform has taken place, $Reform_{jt} = 1$, then $\tau_{it}^k = 0$ and

$$\log(mrpk_{ijt}) = \log(p_t^k)$$

Hence, if $Reform_{jt} = 0$:

$$\log(MRPK_{ijt}) = \log(1 + \tau_{it}^k) + \log(p_t^k) + \mu_i + \eta_t + \epsilon_{it}$$

For firms where $Reform_{jt} = 1$:

$$\log(MRPK_{ijt}) = \log(p_t^k) + \mu_i + \eta_t + \epsilon_{it}.$$

Denote $g(Reform_{jt}, \mathbf{C}_{it})$ to be a function of the time-varying, reform indicator variable and time-varying firm-level characteristics \mathbf{C}_{it} . Then, the difference-in-

differences regression estimates

$$\log(MRPK_{ijt}) = g(Reform_{jt}, \mathbf{C}_{it}) + \mathbf{\Gamma}\mathbf{X}_{it} + \alpha_i + \delta_t + \epsilon_{it}.$$

In this regression, firm fixed effects absorb μ_i , as well as any time invariant industry shocks, and time fixed effects absorb η_t and $\log(p_t^k)$. Idiosyncratic shocks ϵ_{it} are independent of $Reform_{jt}$. Thus, $\hat{g}_i(1, \mathbf{C}_{it})$ is an unbiased estimator of $E(\log(1+\tau_{it}))$ over the pre-period and can be used to predict the average value of $\log(1+\tau_i^k)$ over the pre-period.

Appendix C: Non-Linear Approximation

This appendix describes how we calculate a non-linear approximation of the policies' effects on the treated industries' Solow residual. Following Baqaee and Farhi (2019), we note that a non-linear approximation of the effect of the policies on the Solow residual – given the shocks realized in the economy – is a Reimann sum over the first order approximations of the policies' effects each year. Then, the non-linear approximation of the culmulative effect at time T is

$$\Delta Solow_{I,T} \approx \sum_{t \le T} \sum_{i \in I} \lambda_{it} \Delta \log A_{it} + \sum_{t \le T} \sum_{\substack{i \in I \\ x \in \{K,L,M\}}} \lambda_{it} \ \alpha_i^x \ \frac{\tau_i^x}{1 + \tau_i^x} \ \Delta \log x_{it}, \quad (13)$$

where t indexes a year, and the summation begins in the year of the first policy change. As before, since we did not find the policy had a significant effect on TFPQ (see Table 5), we set $\Delta \log A_{it} = 0$, causing the first term of equation (13) to drop out. We calculate λ_{it} exactly as we did in Section 6, except that we now calculate a separate value for each year, instead of only using the Prowess data from 2000. Similarly, the output elasticities are still given by the production function estimates.

To arrive at a time-varying estimate of the policies' effects on inputs, we use more flexible regressions specifications. For capital, we estimate

$$\log K_{ijtd} = \sum_{s=1}^{5} \beta_{1,s} I_{jt}^{s \ge d} + \beta_{2,s} I_{jt}^{s \ge d} \times I_{i}^{High \ MRPK} + \beta_{3,s} I_{jt}^{s \ge d} \times I_{i}^{High \ MRPL} + \mathbf{\Gamma} \mathbf{X}_{it} + \alpha_{i} + \delta_{t} + \epsilon_{ijt},$$
(14)

where d indexes the number of years since a reform occurred in industry j, and $I_{jt}^{s\geq d}$ in an indicator variable equal to 1 if it has been more than s years since a reform occurred in industry j. Therefore, $\beta_{1,s}$ captures the change in capital that occurs due to the reform between s-1 years after the reform and s years after the reform, and $\beta_{2,s}$ and $\beta_{3,s}$ allow these changes to be heterogeneous for high MRPK and MRPL firms. We allow effects to vary up to 5 years after the policies took place since the effects of the policies appear to plateau after five years (see Figures 1 and 2). Then, to estimate the firm-level change in capital due to the policy in year t, we calculate

$$\widehat{\log K_{ijtd}} = \sum_{s=1}^{5} \hat{\beta}_{1,s} I_{jt}^{d=s} + \hat{\beta}_{2,s} I_{jt}^{d=s} \times I_i^{High\ MRPK} + \hat{\beta}_{3,s} I_{jt}^{d=s} \times I_i^{High\ MRPL},$$

where $I_{jt}^{d=s}$ is an indicator variable equal to 1 if it is s years after an event in industry j and time t. We use an analogous approach to estimate the change in labor by year.

To estimate the baseline wedges in each year, we replace the outcome variable in equation (14) with $\log MRPK_{ijt}$ and $\log MRPL_{ijt}$. Then, under the lower bound identifying assumption that the policy eliminated misallocation, the wedge at time t is the sum of the estimated changes in the wedges that occurred between t and T. For the wedge on capital, after estimating equation (14) with $\log MRPK$ as the outcome variable, this is given by

$$\log \widehat{MRPK}_{ijtd} = \sum_{n=t}^{T} \sum_{s=1}^{5} \hat{\beta}_{1,s} I_{jn}^{d=s} + \hat{\beta}_{2,s} I_{jn}^{d=s} \times I_{i}^{High \ MRPK} + \hat{\beta}_{3,s} I_{jn}^{d=s} \times I_{i}^{High \ MRPL}.$$

The method for identifying the time-varying wedges for labor is analogous.

Appendix Figures

Figure A1: Flow of Foreign Equities



This figure plots the overall amount of foreign equity in Prowess for industries that have deregulated in 2001 (the red line), in 2006 (the green line) or whose regulation did not change during the period 1995–2015 (the blue line). The flows are normalized to one in 2001.



Figure A2: Separate Event Studies for High and Low MRPK Firms

This figure reports the effect of FDI deregulation for high and low MRPK firms separately for physical assets, MRPK, revenues, and the wage bill. The dependent variables are in logs. The reform is normalized to take place in year 1. Each dot is the coefficient on the interaction between being observed t years after the reform and being in a treated industry. The confidence interval is at the 95% level.

Figure A3: Ex-ante MRPK Dispersion by State-level Financial Development



This figure reports a measure of the pre-treatment dispersion of MRPK by state-level financial development. For each state-industry bin, we compute the average MRPK for high and low MRPK firms and take the difference. We then express these differences as a fraction of the total distance (namely we compute: [MRPK_{high}-MRPK_{low})/(MRPK_{high}+MRPK_{low}]) and plot this measure against state-level financial development.

Appendix Tables

Table A1: List of Industries Affected by the 2001 and 2006 Reforms

NIC 5-Digit Industry Classification	Reform Year
Manufacture of 'ayurvedic' or 'unani' pharmaceutical preparation	2001
Manufacture of allopathic pharmaceutical preparations	2001
Manufacture of medical impregnated wadding, gauze, bandages, dressings, surgical gut string etc.	2001
Manufacture of homoeopathic or biochemic pharmaceutical preparations	2001
Manufacture of other pharmaceutical and botanical products n.e.c. like hina powder etc.	2001
Manufacture of rubber types and tubes n.e.c.	2006
Manufacture of essential oils; modification by chemical processes of oils and fats (e.g. by oxidation, polymerization etc.)	2006
Manufacture of various other chemical products	2006
Manufacture of rubber types and tubes for cycles and cycle-rickshaws	2006
Manufacture of distilled, potable, alcoholic beverages such as whisky, brandy, gin, 'mixed drinks' etc.	2006
Coffee curing, roasting, grinding blending etc. and manufacturing of coffee products	2006
Retreading of types; replacing or rebuilding of tread on used pneumatic types	2006
Manufacture of chemical elements and compounds doped for use in electronics	2006
Manufacture of country liquor	2006
Manufacture of matches	2006
Manufacture of rubber plates, sheets, strips, rods, tubes, pipes, hoses and profile -shapes etc.	2006
Distilling, rectifying and blending of spirits	2006
Manufacture of bidi	2006
Manufacture of catechu(katha) and chewing lime	2006
Stemming and redrying of tobacco	2006
Manufacture of other rubber products n.e.c.	2006
Manufacture of rubber contraceptives	2006
Manufacture of other tobacco products including chewing tobacco n.e.c.	2006
Manufacture of pan masala and related products.	2006

This table lists 5-digit NIC industries that changed to automatic foreign investment approval for investments up to (at least) 51% of a firm's capital and the year that the policy reform took place.

Dependent Variable	Land	Plants and Equipment	Infrastructure	Other
	(1)	(2)	(3)	(4)
$Reform_{jt} \times I_i^{High MRPK}$	-0.022^{**} (0.009)	0.038^{***} (0.009)	-0.001 (0.003)	-0.015 (0.010)
$Reform_{jt}$	$0.005 \\ (0.014)$	-0.009 (0.009)	-0.004 (0.005)	$\begin{array}{c} 0.007 \\ (0.009) \end{array}$
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	59,218	59,218	59,218	$59,\!218$

Table A2: Composition of Change in Capital

This table reports estimates of the heterogeneous effects of foreign capital liberalization reforms on high and low MRPK firms in the Prowess dataset (equation (5)). All dependent variables are the share of capital in a category. Firms are observed between 1995 and 2015. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is calculated as Revenue/K. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10, 5, and 1% statistical significance respectively.

Dependent Variable	Total Capital	Variance(MRPK)			
	(1)	(2)	(3)	(4)	
		Panel A: P	rowess		
Share of (4-digit) Industry Treated	0.470^{*} (0.267)	-0.469^{**} (0.219)	-1.247^{***} (0.310)	-1.116^{***} (0.352)	
Fixed Effects Industry (4-digit) Industry (2-digit) × Year	\checkmark	\checkmark	\checkmark	\checkmark	
Controls Output Tariff Input Tariff Dereservation Nb firms Observations	 2,289	 1,789		✓ ✓ ✓ 1,789	
		Panel B:	ASI		
Share of (4-digit) Industry Treated	0.782^{***} (0.276)	-0.615^{***} (0.221)	-0.668^{***} (0.235)	-0.657^{***} (0.243)	
Fixed Effects Industry (4-digit) Industry (2-digit) × Year	\checkmark	\checkmark	\checkmark	\checkmark	
Controls Output Tariff Input Tariff Dereservation Nb firms	 			\checkmark \checkmark \checkmark	
Observations	2,139	$2,\!101$	2,101	2,101	

Table A3: Industry-Level Results in Prowess and the ASI

This table reports the effects of the reform on total 4-digit industry-level log capital (column 1) and the 4-digit industry-level log variance of MRPK in Prowess (Panel A) an the ASI (Panel B). Standard errors are two-way clustered at the 4-digit industry and year level. ***, **, * indicate statistical significance at the 1%, 5%, and 10% levels, respectively.

Dependent Variable	Revenues	Capital	Wages	MRPK
	(1)	(2)	(3)	(4)
	Panel A: 1	995-1997	Pre-treatm	ent Period
$Reform_{jt} \times I_i^{High MRPK}$	0.207^{***} (0.066)	0.427^{***} (0.095)	0.094 (0.058)	-0.215^{***} (0.059)
$Reform_{jt}$	0.054 (0.091)	0.152 (0.125)	0.162 (0.120)	-0.072 (0.095)
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$45,\!602$	46,891	$46,\!179$	44,579
	<u>Panel B: 1</u>	995-1998	Pre-treatm	ent Period
$Reform_{jt} \times I_i^{High MRPK}$	0.203^{***} (0.066)	0.470^{***} (0.081)	0.224^{***} (0.058)	-0.285^{***} (0.077)
$Reform_{jt}$	(0.000) (0.020) (0.106)	(0.031) 0.075 (0.112)	(0.058) 0.063 (0.103)	(0.017) -0.019 (0.106)
Fixed Effects	· · ·	. ,	. ,	. ,
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	v v	\checkmark	↓	↓
Size \times Year	v v	• •	·	v
Observations	50,092	5 1,526	5 0,730	48,989
	Par	nel C: Only	y 2006 Ref	orm
Deferment High MRPK	0.105	0 495***	0.940***	0.966
$Reform_{jt} \times I_i^{High MRPK}$	$0.195 \\ (0.123)$	0.425^{***} (0.091)	$\begin{array}{c} 0.249^{***} \\ (0.089) \end{array}$	-0.266 (0.163)
$Reform_{jt}$	-0.047 (0.204)	-0.077 (0.115)	-0.085 (0.107)	$0.102 \\ (0.149)$
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	58,391	60,096	59,162	57,017

Table A4: Heterogeneous Effects of Foreign Capital Liberalization and Mean Reversion

This table provides evidence that the results in Table 4 are not driven by mean reversion. Firms are classified as high MRPK if their average MRPK in a pre-treatment period is above the 4-digit industry median. In Panel A, the pre-treatment period is defined as 1995-1997. In Panel B, it is 1995-1998. In Panel C, the pre-treatment period is 1995-2000, but the treatment effect is only reported for the 2006 reform. In Panel C, the regressions control separately for being treated by the 2001 reform and its interaction with high MRPK. MRPK is calculated as *Revenue/K*. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Revenues	Capital	Wages	MRPK
	(1)	(2)	(3)	(4)
	Panel A	: (5-digit)	Industry-l	oy-Year
$Reform_{jt} \times I_i^{High MRPK}$	0.371^{***} (0.067)	0.634^{***} (0.076)	0.376^{***} (0.060)	-0.280^{**} (0.112)
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	1	\checkmark	\checkmark
Size \times Year	• •	, ,	✓	• √
5-digit Industry \times Year	v	√	√	↓
Observations	58,372	60,062	59,139	5 6,999
	Panel B	: (2-digit)	Industry-l	oy-Year
$Reform_{jt} \times I_i^{High MRPK}$	0.248***	0.542***	0.284***	-0.308***
	(0.069)	(0.067)	(0.055)	(0.106)
$Reform_{it}$	-0.137	-0.121	-0.105	0.029
<i>.</i>	(0.145)	(0.121)	(0.082)	(0.131)
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
2-digit Industry \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	58,372	60,062	$59,\!139$	$56,\!999$
	Pa	anel C: Sta	ate-by-Yea	r
$Reform_{jt} \times I_i^{High MRPK}$	0.237***	0.510***	0.271***	-0.299**
	(0.083)	(0.073)	(0.071)	(0.112)
$Reform_{it}$	-0.040	0.033	0.006	-0.031
10.joini _{jt}	(0.133)	(0.033) (0.089)	(0.095)	(0.106)
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
State \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	58,319	60,009	59,101	56,947

Table A5: Robustness to High-Dimensional Fixed Effects

This table reports estimates of the heterogeneous effects of the liberalization reforms on high MRPK firms in the Provess dataset (equation (5)). All dependent variables are in logs. Firms are observed between 1995 and 2015. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is calculated as Revenue/K. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	le Revenues		Cap	Capital		Wages		MRPK	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
$Reform_{jt} \times I_i^{High MRPK}$	0.261***	0.230***	0.654***	0.483***	0.370***	0.284***	-0.429***	-0.279*	
	(0.074)	(0.077)	(0.044)	(0.114)	(0.065)	(0.058)	(0.077)	(0.141)	
Reform _{it}	0.061	-0.025	0.038	0.033	0.056	0.003	0.066	-0.010	
	(0.100)	(0.125)	(0.079)	(0.070)	(0.106)	(0.097)	(0.122)	(0.105)	
Fixed Effects									
Firm	\checkmark								
Firm Age	\checkmark								
Size \times Year	\checkmark								
Dereservation Controls		\checkmark		\checkmark		\checkmark		\checkmark	
Observations	26,111	$58,\!391$	26,986	60,096	$26,\!539$	59,162	$25,\!406$	$57,\!017$	
Sample	Restricted	All	Restricted	All	Restricted	All	Restricted	All	

Table A6:	Robustness	to	Accounting	for	Dereservation
-----------	------------	---------------------	------------	-----	---------------

This table reports estimates of the heterogeneous effects of foreign capital liberalization reforms on high and low MRPK firms in the Provess dataset (equation (5)), accounting for dereservation policies. Firms are observed between 1995 and 2015. In odd columns, we restrict the sample to firms in industries not affected by a dereservation policy after 2000 (i.e. a change in regulation specific to small and medium size firms). Data on dereservation events come from Boehm, Dhingra, and Morrow (2019). In even columns, we include the whole sample but interact $I_i^{High MRPK}$ with an indicator variable *Dereservation*_{jt} that is equal to 1 after the industry has been dereserved. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the industry median. MRPK is approximated as *Revenue/K*. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10, 5, and 1% statistical significance respectively.

Table A7: Effect of	Foreign	Canital	Liberalization	Controlling	r for	Tariffs
	rorongin	Capital	LIDCI anZaulon,	Controning	, 101	rarmo

Dependent Variable	Reve	nues	Capital		Wages		MRPK	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
$Reform_{jt} \times I_i^{High MRPK}$	0.209^{***} (0.076)	0.189^{*} (0.097)	0.503^{***} (0.081)	0.493^{***} (0.062)	0.272^{***} (0.058)	0.297^{***} (0.071)	-0.319^{***} (0.100)	-0.283^{**} (0.133)
$Reform_{jt}$	$\begin{array}{c} 0.042 \\ (0.130) \end{array}$	$\begin{array}{c} 0.096 \\ (0.231) \end{array}$	$0.165 \\ (0.106)$	$0.148 \\ (0.143)$	$\begin{array}{c} 0.118 \\ (0.098) \end{array}$	$\begin{array}{c} 0.114 \\ (0.129) \end{array}$	-0.090 (0.094)	-0.044 (0.121)
Tariff Controls								
Output Tariffs	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Input Tariffs	—	\checkmark	—	\checkmark	—	\checkmark	—	\checkmark
Fixed Effects								
Firm	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations	54,280	$43,\!458$	55,936	44,899	55,100	44,312	53,060	42,713

This table reports estimates of the heterogeneous effects of foreign capital liberalization on high and low pretreatment MRPK firms (equation (5)) over the period 1995-2015, controlling for the effects of tariff policies and allowing those tariff policies to have differential effects by high and low MRPK. All dependent variables are in logs. $Reform_{jt}$ is an indicator variable equal to one if the industry has liberalized access to international capital market. Firms are classified as high MRPK if their average MRPK in the pre–treatment period from 1995-2000 is above the 4-digit industry median. Tariff data from 1995-2010 are constructed following Goldberg, Khandelwal, Pavcnik, and Topalova (2010), and tariff levels are coded at the 2010 level from 2010-2015. Output tariff controls are the average tariff on an industry and its interaction with $I_i^{High MRPK}$. Input tariff controls are the average tariff on the inputs used by an industry and its interaction with $I_i^{High MRPK}$. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Revenues	Capital	Wages	MRPK
	(1)	(2)	(3)	(4)
	Panel A: W	insorized 50	% Across In	dustries
$Reform_{jt} \times I_i^{High MRPK}$	0.118^{**} (0.053)	0.495^{***} (0.088)	0.195^{***} (0.046)	-0.348^{***} (0.069)
$Reform_{jt}$	$0.026 \\ (0.087)$	-0.003 (0.077)	$\begin{array}{c} 0.019 \\ (0.082) \end{array}$	$\begin{array}{c} 0.003 \ (0.095) \end{array}$
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	58,391	60,096	59,162	57,017
	Panel B: W	inzorized 5%	% Within Ir	ndustries
$Reform_{jt} \times I_i^{High MRPK}$	0.128^{**} (0.050)	0.496^{***} (0.085)	0.188^{***} (0.049)	-0.368^{***} (0.064)
Reform _{it}	0.014	-0.003	0.037	0.009
$nejorm_{jt}$	(0.014)	(0.078)	(0.031)	(0.096)
Fixed Effects				
Firm	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark
Observations	58,391	60,096	59,162	57,017

Table A8: Results after Winsorizing the Data

This table reports estimates of the heterogeneous effects of foreign capital liberalization on capital constrained and unconstrained firms after winsorizing the top and bottom 5% of the sample for each outcome. In Panel A, the sample is winsorized *across* industries, while in Panel B, the sample is winsorized *within* 2-digit industries. All dependent variables are in logs. Firms are observed between 1995 and 2015. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is calculated as Revenue/K. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Number of Exits		Number of Entrants
	(1)	(2)	(3)
$Reform_{jt}$	$0.086 \\ (0.085)$	$0.045 \\ (0.039)$	-0.031 (0.030)
$Reform_{jt} \times I_i^{High MRPK}$		-0.013 (0.021)	
Fixed Effects Ind-5 digit Year Observations	\checkmark \checkmark 6,575	√ √ 11,673	$\checkmark \\ \checkmark \\ 6,575$

Table A9: Effects of Foreign Capital Liberalization on Firm Exit and Entry

This table estimates the effect of the foreign capital liberalization on firm exit and entry in the Prowess data. In columns 1 and 3, an observation is a 5-digit industry-year cell. In column 2, it is a 5-digit industry-year-MRPK category cell. A firm is counted as exiting in a year if it is not observed in the data in that year and does not reenter the data in a later year. A firm is counted as entering in a year if that is the year of the firm's incorporation. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. In column 2, MRPK is calculated as Revenue/K. Standard errors are two-way clustered at the 4-digit industry and year level.

Dependent Variable	Revenues		Capital		Wages		MRPK	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
$Reform_{jt}$	$0.106 \\ (0.094)$	-0.025 (0.127)	0.308^{**} (0.115)	$0.008 \\ (0.086)$	$0.154 \\ (0.095)$	-0.007 (0.094)	-0.179^{**} (0.084)	0.011 (0.099)
$Reform_{jt} \times I_i^{High MRPK}$		$\begin{array}{c} 0.227^{***} \\ (0.077) \end{array}$		$\begin{array}{c} 0.525^{***} \\ (0.076) \end{array}$		$\begin{array}{c} 0.279^{***} \\ (0.058) \end{array}$		-0.323^{***} (0.106)
Fixed Effects								
Firm	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations	58,391	58,391	60,096	60,096	59,162	59,162	57,017	57,017

Table A10: Effects of Capital Liberalization, Accounting for Spillovers

This table reports estimates of the heterogeneous effects of foreign capital liberalization on high and low MRPK firms, controlling for spillovers through the input-output matrix. All dependent variables are in logs. Firms are observed between 1995 and 2015. The regressions include controls for $Upstream_{jt}$, which measures the composite reform shock to an industry from upstream industries, and $Downstream_{jt}$, which measures the composite reform shock from downstream industries. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is calculated as Revenue/K. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Revenues	Capital	Wages	MRPK	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_i^{High MRPK}$	$0.057 \\ (0.063)$	0.508^{***} (0.061)	0.159^{***} (0.043)	-0.470^{***} (0.105)	
$Reform_{jt}$	$\begin{array}{c} 0.079 \\ (0.128) \end{array}$	$0.022 \\ (0.092)$	$0.081 \\ (0.106)$	$0.099 \\ (0.098)$	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	$60,\!275$	62,042	$61,\!075$	$58,\!885$	

Table A11: Robustness to More Parsimonious Controls

This table reports estimates of the effect of foreign capital liberalization on high and low pre-treatment MRPK firms (equation (5)) over the period 1995–2015. All dependent variables are in logs. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is estimated with the Revenue/K method. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Revenues	Capital	Wages	MRPK	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_i^{High MRPK}$	0.240^{***} (0.072)	0.561^{***} (0.067)	$\begin{array}{c} 0.268^{***} \\ (0.055) \end{array}$	-0.354^{***} (0.100)	
$Reform_{jt}$	-0.026 (0.114)	-0.004 (0.077)	$\begin{array}{c} 0.021 \\ (0.094) \end{array}$	$\begin{array}{c} 0.021 \\ (0.112) \end{array}$	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	$62,\!924$	$65,\!393$	$63,\!999$	$61,\!342$	

Table A12: Robustness to Using Data Without Filters

This table reports estimates of the effect of foreign capital liberalization on high and low pre-treatment MRPK firms (equation (5)) over the period 1995–2015 when we do not remove firms with large contractions in sales from the data. All dependent variables are in logs. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is estimated with the *Revenue/K* method. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Revenues	Capital	Wages	MRPK	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_i^{High MRPK}$	0.201^{*} (0.100)	0.472^{***} (0.098)	0.230^{***} (0.074)	-0.256^{**} (0.105)	
$Reform_{jt} \times Revenue \ Beta_i$	0.003 (0.015)	0.027 (0.020)	-0.003 (0.019)	-0.025^{*} (0.014)	
$Reform_{jt}$	0.032 (0.119)	0.064 (0.086)	(0.081) (0.089)	-0.018 (0.111)	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	50,087	$51,\!456$	50,740	48,912	

Table A13: Effects of Foreign Capital Liberalization, Accounting for Firm-Risk

This table reports estimates of the heterogeneous effects of foreign capital liberalization on high and low MRPK firms, controlling for the firm's pre-treatment risk and its interaction with the reform. All dependent variables are in logs. Firms are observed between 1995 and 2015. *Revenue Beta*_i is computed as the correlation between firm revenue growth up to 2000 and the average revenue growth in the economy. Firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median. MRPK is calculated as *Revenue/K*. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Table A14.	Pohystroga t	o Alternative	MDDK	Cut Offe
Table A14:	Robustness t	o Alternative	MAPA	Out-Ons

Dependent Variable	Revenues		Capital		Wages		MRPK	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
$Reform_{jt}$	0.019 (0.087)	-0.112 (0.146)	0.139^{*} (0.080)	-0.178 (0.105)	$0.095 \\ (0.084)$	-0.064 (0.120)	-0.087 (0.103)	0.103 (0.133)
$Reform_{jt} \times I_{jt}^{High MRPK-Mean}$	0.255^{**} (0.110)		0.473^{***} (0.120)		$\begin{array}{c} 0.215^{**} \\ (0.089) \end{array}$		-0.257^{***} (0.071)	
$Reform_{jt} \times I_{jt}^{MRPK}$ Tercile =2		$\begin{array}{c} 0.245 \\ (0.150) \end{array}$		$\begin{array}{c} 0.631^{***} \\ (0.099) \end{array}$		$\begin{array}{c} 0.310^{***} \\ (0.103) \end{array}$		-0.372^{**} (0.174)
$Reform_{jt} \times I_{jt}^{MRPK \ Tercil=3}$		$\begin{array}{c} 0.402^{***} \\ (0.090) \end{array}$		$\begin{array}{c} 0.801^{***} \\ (0.142) \end{array}$		$\begin{array}{c} 0.380^{***} \\ (0.076) \end{array}$		-0.447^{***} (0.142)
Fixed Effects								
Firm	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations	$62,\!924$	62,924	$65,\!393$	$65,\!393$	$63,\!999$	63,999	$61,\!342$	$61,\!342$

This table reports estimates of the heterogeneous effects of foreign capital liberalization on high and low MRPK firms. All dependent variables are in logs. Firms are observed between 1995 and 2015. In even columns, firms are classified as high MRPK if their average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry mean (instead of median). In odd columns, firms are classified as belonging to the second or third tercile of average MRPK distribution within their 4 digit industry. MRPK is calculated as Revenue/K. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.

Dependent Variable	Capital	Wages	MRPK	MRPL	
	(1)	(2)	(3)	(4)	
$Reform_{jt} \times I_i^{High MRPK}$	0.540***	0.242***	-0.366***	-0.129	
	(0.081)	(0.061)	(0.108)	(0.080)	
$Reform_{jt} \times I_i^{High MRPL}$	0.323***	0.114*	-0.246***	-0.330***	
v <u>j</u>	(0.107)	(0.060)	(0.059)	(0.069)	
$Reform_{it}$	-0.129***	0.008	0.144	0.201*	
<i>u</i> j-	(0.046)	(0.109)	(0.112)	(0.115)	
Fixed Effects					
Firm	\checkmark	\checkmark	\checkmark	\checkmark	
Firm Age	\checkmark	\checkmark	\checkmark	\checkmark	
Size \times Year	\checkmark	\checkmark	\checkmark	\checkmark	
Observations	59,802	$58,\!898$	$56,\!557$	46,064	

Table A15: Regression Estimates Used to Estimate the Effect of the Policy on the Solow Residual

This table reports the difference-in-differences estimates used to estimate the policy's effects on treated industries' Solow residual. All dependent variables are in logs. Firms are observed between 1995 and 2015. $I_i^{High, MRPK}$ is coded as 1 if a firm's average MRPK in the pre-treatment period from 1995-2000 is above the 4-digit industry median, where MRPK is calculated using *Revenue/K*. $I_i^{High, MRPL}$ is defined analogously for labor. Size×Year are quartile fixed effects for firms' average pre-treatment capital interacted with year fixed effects. Standard errors are two-way clustered at the 4-digit industry and year level. *, **, and *** denote 10%, 5%, and 1% statistical significance respectively.